# DOCUMENTO DE TRABAJO

DEPARTAMENTO DE ECONOMIA Pontificia Universidad Católica del Perú EPARTAMENT N° 523 FPART **NATA** Pontificia DEPAR )NOMÍ IMPACT OF MONETARY POLICY SHOCKS IN THE Pontificia PERUVIAN ECONOMY **OVER TIME** Pontificia a del Perí ΟΛΙΔ Gabriel Rodriguez ra del Períi y Flavio Pérez Rojo Pontificia Universidad Católica del Per PARTAMENT Pontificia Universidad Católica del Perú

DECON



# DOCUMENTO DE TRABAJO N° 523

Impact of Monetary Policy Shocks in the Peruvian Economy Over Time Flavio Pérez Rojo y Gabriel Rodríguez

Agosto, 2023



DOCUMENTO DE TRABAJO 523 http://doi.org/10.18800/2079-8474.0523 **Impact of Monetary Policy Shocks in the Peruvian Economy Over Time** Documento de Trabajo 523

@ Flavio Pérez Rojo, Gabriel Rodríguez

Editado e Impreso: © Departamento de Economía – Pontificia Universidad Católica del Perú Av. Universitaria 1801, Lima 32 – Perú. Teléfono: (51-1) 626-2000 anexos 4950 - 4951 <u>econo@pucp.edu.pe</u> http://departamento.pucp.edu.pe/economia/publicaciones/documentos-de-trabajo/

Encargada de la Serie: Janina V. León Castillo Departamento de Economía – Pontificia Universidad Católica del Perú jaleon@pucp.edu.pe Primera edición – Agosto, 2023

ISSN 2079-8474 (En línea)

# Impact of Monetary Policy Shocks in the Peruvian Economy Over $$\mathrm{Time}^*$$

Flavio Pérez Rojo<sup>†</sup> Pontificia Universidad Católica del Perú Banco Central de Reserva del Perú Gabriel Rodríguez<sup>‡</sup> Pontificia Universidad Católica del Perú

This Version: August 21, 2023

#### Abstract

We investigate the evolution of the impact of monetary policy (MP) shocks in Peru in 1996Q1-2018Q2 using a set of time-varying parameter vector autoregressive models with stochastic volatility (TVP-VAR-SV), as proposed by Chan and Eisenstat (2018). The main results are: (i) the volatilities, intercepts, and contemporaneous coefficients change more gradually than VAR coefficients over time; (ii) the volatility of MP shocks falls from 4% to 0.3% on average during the Inflation Targeting (IT) regime; (iii) in the long run, a contractionary MP shock decreases both gross domestic product (GDP) growth and inflation by 0.28% and 0.1%, respectively; (iv) the interest rate reacts faster to aggregate supply shocks than to both aggregate demand shocks and exchange rate shocks; (v) under the pre-IT regime, MP shocks explain almost 20%, 10%, and 85% of the uncertainty in GDP growth, inflation, and the interest rate, respectively; and under the IT regime, all these percentages shrink to 1-2%. The sensitivity analysis confirms the robustness of the main results across various prior specifications, measures of external and domestic variables, and recursive identifications. In general, the results show that MP has contributed to diminishing macroeconomic volatility in Peru.

Classification JEL: C32, E32, E51, E52.

Keywords: Time-Varying Parameter VAR, Stochastic Volatility, Marginal Likelihood, Deviance Information Criterion, Monetary Policy, Peru.

<sup>\*</sup>This document is drawn from the Thesis of Flavio Pérez Rojo at the Faculty of Social Sciences of the Pontificia Universidad Católica del Perú (PUCP). We are grateful for the comments by Paul Castillo B. (Central Reserve Bank of Peru and PUCP). The views expressed in this paper are those of the authors and do not necessarily reflect the positions of the Central Reserve Bank of Peru. Any remaining errors are our responsibility.

<sup>&</sup>lt;sup>†</sup>E-Mail Address: flavio.perez@pucp.edu.pe and flavio.perez@bcrp.gob.pe.

<sup>&</sup>lt;sup>‡</sup>Autor de Correspondencia: Gabriel Rodríguez, Departamento de Economía, Pontificia Universidad Católica del Perú, 1801 Universitaria Avenue, Lima 32, Lima, Perú, Teléfono: +511-626-2000 (4998), E-Mail Address:: gabriel.rodriguez@pucp.edu.pe, ORCID ID: https://orcid.org/0000-0003-1174-9642.

# Impacto de los Choques de Política Monetaria en la Economía Peruana a lo Largo del Tiempo<sup>\*</sup>

Flavio Pérez Rojo<sup>†</sup> Pontificia Universidad Católica del Perú Banco Central de Reserva del Perú Gabriel Rodríguez<sup>‡</sup> Pontificia Universidad Católica del Perú

Versión: August 21, 2023

#### Resumen

Investigamos la evolución del impacto de los choques de política monetaria (MP) en Perú en 1996Q1-2018Q2 utilizando un conjunto de modelos de vectores autorregresivos de parámetros variantes en el tiempo con volatilidad estocástica (TVP-VAR-SV), según lo propuesto por Chan y Eisenstat (2018). Los principales resultados son: (i) las volatilidades, los interceptos y los coeficientes contemporáneos cambian más gradualmente que los coeficientes VAR a lo largo del tiempo; (ii) la volatilidad de los choques de MP cae de 4% a 0.3% en promedio durante el régimen de Metas de Inflación (IT); (iii) en el largo plazo, un choque de MP contractivo reduce tanto el crecimiento del producto interno bruto (PIB) como la inflación en 0.28% y 0.1%, respectivamente; (iv) la tasa de interés reacciona más rápido a los choques de oferta agregada que a los choques de demanda agregada y a los choques de tipo de cambio; (v) bajo el régimen pre-IT, los choques de interés, respectivamente; y bajo el régimen de IT, todos estos porcentajes se reducen al 1-2%. El análisis de sensibilidad confirma la solidez de los principales resultados a través de varias especificaciones de priors, medidas de variables externas e internas e identificaciones recursivas. En general, los resultados muestran que la MP ha contribuido a disminuir la volatilidad macroeconómica en el Perú.

#### Clasificación JEL: C32, E32, E51, E52.

**Palabras Clave:** VAR con parámetros Cambiantes, Volatilidad Estocástica, Verosimilitud Marginal, Criterio de Información de Desviación, Política Monetaria, Perú.

<sup>\*</sup>Este documento se basa en la Tesis de Licenciatura de Flavio Pérez Rojo, Facultad de Ciencias Sociales de la Pontificia Universidad Católica del Perú (PUCP). Agradecemos los valiosos comentarios de Paul Castillo B. (Banco Central de Reserva del Perú y PUCP). Las opiniones expresadas en este documento son de los autores y no necesariamente reflejan la posición del Banco Central de Reserva del Perú. Cualquier error remanente es nuestra responsabilidad.

<sup>&</sup>lt;sup>†</sup>Correo Electrónico: flavio.perez@pucp.edu.pe and flavio.perez@bcrp.gob.pe.

<sup>&</sup>lt;sup>‡</sup>Autor de Correspondencia: Gabriel Rodríguez, Departamento de Economía, Pontificia Universidad Católica del Perú, Avenida Universitaria 1801, Lima 32, Lima, Perú, Teléfono: +511-626-2000 (4998), Correo Electrónico: gabriel.rodriguez@pucp.edu.pe, ORCID ID: https://orcid.org/0000-0003-1174-9642.

#### 1 Introduction

Papers like Primiceri (2005), Canova et al. (2007), and Nakajima (2011) analyze the time-varying impact of monetary policy (MP) shocks on inflation, unemployment, and GDP growth in developed economies. However, there is still a gap in the literature for emerging market economies (EMEs). The analysis of these economies is complex, as external shocks are the main source of uncertainty (see, for instance, Drechsel and Tenreyro (2018), Fernández et al. (2018), Ojeda Cunya and Rodríguez (2022), Chávez and Rodríguez (2023), and Rodríguez et al. (2023b)). Research suggests that the potential of MP for stabilizing macroeconomic dynamics increases as it strengthens its impact on domestic variables and its reaction to external shocks. This article discusses the evolving impact of MP shocks on inflation and GDP growth, as well as the MP response to various shocks in Peru.

Peru is a unique case among EMEs, as it has experienced numerous reforms over the past three decades. Until the 1980s, the Peruvian economy experienced high uncertainty due to hyperinflation and a debt crisis. In the early 1990s, the Fujimori administration introduced a stabilization program, deregulated markets, and reduced government involvement in the economy. Price controls were abandoned, capital markets were liberalized, and the FX market was unified. The Central Reserve Bank of Peru (BCRP) was constitutionally granted operational independence, thus putting an end to inflationary fiscal financing. Since then, MP design and implementation have undergone many changes. Initially, to address hyperinflation, price stability was considered the BCRP's sole objective, with the monetary base as the operative instrument, while interest rates and the exchange rate were freely determined by the market.

However, the Asian-Russian crisis led to a dollar outflow, a credit crunch, and deflation, which lasted until 2001. The correlation between the monetary base and inflation also declined, weakening the monetary aggregate regime. As a result, the BCRP adopted an Inflation Targeting (IT) regime in January 2002, with a 1.5%-3.5% target band to anchor inflation expectations and the BCRP's current accounts as the operational instrument. To improve MP signaling, the BCRP introduced a reference interest rate as the primary instrument in September 2003. Since then, various alternative tools, such as FX swaps and reverse repos, have been implemented. In February 2007, the BCRP adjusted the target band to 1%-3%, aiming to increase flexibility in addressing adverse shocks and aligning domestic inflation with that of major developed economies.

These modifications in the instruments lead to potential changes in the transmission of MP shocks, which can be modeled using Time-Varying Parameter Vector Autoregression (TVP-VAR) models, as detailed by Cogley and Sargent (2001) and Boivin and Giannoni (2006). Additionally, some literature suggests that macroeconomic variable volatilities change over time (e.g., Sims and Zha (2006) and Benati (2008)). For instance, during episodes like the Global Financial Crisis (GFC) and the Gulf War, foreign variables such as the S&P GSCI and export price indices typically reach peak local volatility. These changes in the volatility influence MP design, as policymakers assess how to mitigate the impact of diverse shocks on the business cycle. While there are several options for modeling heteroscedasticity in VAR models, the literature mostly agrees that stochastic volatility (SV) outperforms alternative specifications (e.g., Clark (2011) and Clark and Ravazzolo (2015)). Consequently, we employ VAR models that incorporate both TVP and SV, known as TVP-VAR-SV models.

This paper analyzes six recursively identified variables for the period 1996Q1-2018Q2: S&P GSCI growth, exchange rate growth, GDP growth, inflation, money growth, and interest rate. The results indicate that all volatilities and most parameters change over time. Notably, the volatilities of aggregate demand (AD), aggregate supply (AS), money supply (MS), and MP shocks decrease after IT adoption, while the volatilities of external and exchange rate (ER) shocks decline after the

GFC. Moreover, we estimate that a 1% increase in MP shocks reduces GDP growth and inflation by a maximum of 0.28% and 0.1%, respectively. The interest rate impulse response functions (IRFs) confirm the BCRP's anti-inflationary bias, as reactions to AS shocks are stronger than to AD shocks. Forecast error variance decompositions (FEVDs) reveal that MP shocks are only relevant during the pre-IT regime, while external shocks are the primary driver for much of the sample. Similarly, historical decompositions (HDs) show that MP shocks are only relevant during the pre-IT regime.

The paper is organized as follows: Section 2 summarizes the empirical literature that uses TVP-VAR-SV models to study the effect of MP shocks on domestic variables. Section 3 presents the TVP-VAR-SV model and its variants, as proposed by Chan and Eisenstat (2018). Section 4 describes the data and hyperparameters and discusses the empirical results. Section 5 examines robustness exercises. Section 6 concludes.

#### 2 Literature Review

Since Sims (1980), VAR models have been widely used to analyze the impact of MP shocks. For example, Gordon and Leeper (1994), Bernanke and Mihov (1998), and Christiano et al. (1999) discuss the use of VAR models to analyze MP transmission mechanisms. Although these papers vary in their identification strategies for MP shocks, the results generally indicate that an increase in such shocks diminishes both GDP and prices, with the latter typically reacting with a lag. Moreover, these articles assume that parameters and volatilities are time-invariant; however, given that economic series continuously change and their interrelations evolve, this assumption might be inappropriate. Consequently, macroeconometric literature has addressed this gap by proposing models that assume changes in transmission mechanisms, involving modifications in parameters and volatilities.

One of the first papers to empirically analyze the time-varying dynamics between interest rates, inflation, and the unemployment rate in the US was Cogley and Sargent (2001). They proposed a TVP-VAR model with short-run restrictions and found that the response of interest rates to AS shocks increased during the 1990s, coinciding with greater Fed activism. However, Sims (2001) and Stock (2001) emphasize that homoscedastic variance overestimates time-varying parameters. Using similar data, Cogley and Sargent (2005) employed a TVP-VAR-SV model and demonstrated that the reaction of interest rates to AS shocks increased over time and that the volatilities of interest rates and inflation were time-varying, reaching a peak in the 1980s, coinciding with the beginning of a new MP regime in the US.

Primiceri (2005) extended Cogley and Sargent (2005) by incorporating changes in the matrix of contemporary effects and showed that a contractionary MP shock reduced inflation and increased unemployment in the long run, but these effects were time-invariant. Specifically, a 1% increase in the interest rate reduced inflation by 0.1% in the long run. In contrast, he found that the response of the interest rate to inflation and unemployment shocks increased over time. Finally, he discovered that interest rate volatility explained high-inflation and unemployment episodes during the 1970s.

Using a multivariate regime-switching model, Sims and Zha (2006) analyzed the dynamics between a commodity price index, M2, the Fed funds rate, real GDP, the Consumer Price Index (CPI), and the unemployment rate from 1959M1 to 2003M3. They found that the best fit was achieved assuming constant coefficients but changing variances; however, considering changing coefficients, three MP regimes were identified, one of which showed that MP reduced inflation in the early 1980s. Furthermore, they estimated that a 1% contractionary MP shock reduced GDP by 0.1%, but this shock did not have a statistically significant effect on inflation. Imposing sign restrictions on the methodology proposed by Primiceri (2005), Canova et al. (2007) study the time-varying dynamics between GDP growth, inflation, the interest rate, and a monetary aggregate for the US, the Eurozone, and the UK from 1959Q1 to 2004Q4. For the US and the Eurozone, they find that the volatility of MP shocks explains 36% and 25% of the volatility of GDP growth and inflation, respectively; for the UK, these percentages are around 11%. Moreover, Canova et al. (2007) estimate that MP shocks have a time-varying impact on inflation for all economies. In a similar vein, Benati (2008) models the macroeconomic dynamics in the UK and argues that the parameters of the interest rate equation change over time, allowing the identification of several MP regimes; and calculates that MP shocks have a time-varying impact on inflation, GDP growth, and money.

Using a mixture innovation approach to estimate whether, where, when, and how parameters change, Koop et al. (2009) study the time-varying dynamics between the interest rate, inflation, and the unemployment rate from 1953Q1 to 2006Q2. The authors show that MP shocks have a time-varying effect on unemployment and inflation. For instance, they estimate a price puzzle in 1975, in a context of high inflation and interest rates; and show that a 1% contractionary MP shock reduced inflation by 0.02% as a minimum in 2006.

For Japan, Nakajima (2011) incorporates the interest rate zero-lower bound and finds that GDP growth and inflation are invariant to MP shocks during periods when the interest rate reaches the zero lower bound. For the Czech Republic, Franta et al. (2013) use sign restrictions and estimate that MP shocks have a time-varying effect on GDP and the CPI. Specifically, at the beginning of the sample, a 1% MP shock diminishes GDP and the CPI by 2.5% and 1%, respectively. In contrast, by the end of the sample these percentages fall to 1.5% and 0.7%, respectively.

For Malaysia from 1981Q1 to 2010Q4, Bittencourt et al. (2016) emphasize that the financial liberalization of 1987 generated a theoretically consistent reaction of inflation to MP shocks. For instance, in 2010, a 1% increase in the interest rate reduces inflation by 0.3%, while in 1986, inflation is invariant. For China from 1996M1 to 2014M12, Wu and Wei (2016) calculate that the response of inflation to MP shocks varies over time and by instrument; i.e., interest rate shocks have a higher impact on inflation than monetary aggregate shocks.

Based on their analysis of the US interest rate, GDP growth, and inflation from 1954Q3 to 2014Q4, Chan and Eisenstat (2018) estimate a set of TVP-VAR-SV models and compare their performance using two Bayesian criteria: the deviance information criterion (DIC) and the log marginal likelihood (Log-ML). Both criteria suggest that incorporating SV improves the fit. As a result, the authors noted a trend of drifting volatilities for all shocks over time, a phenomenon they identified as the Great Moderation, evidenced by a decrease in interest rate and inflation volatilities during the 1980s. Additionally, they find that the impact of MP shocks on GDP growth and inflation varies over time. For example, the benchmark TVP-VAR-SV model indicates that a 1% increase in the interest rate would reduce inflation and GDP growth by 0.03% and 0.35%, respectively, as a minimum.

In Peru, most research has focused on the effect of MP shocks on various macroeconomic variables, primarily utilizing static VAR models. For instance, Quispe (2000) examined the influence of different monetary aggregates (such as the monetary base and banknotes), the interest rate, and the exchange rate on inflation from 1980 to 1998. He concludes that banknotes account for 30% of the inflation variance in the long term, and a 1% MP shock (based on banknotes) would decrease inflation by 0.5% after one year. In a subsequent study, Quispe (2001) expands the range of monetary aggregates and deduces that the monetary base explains 30% of the inflation variance, and a 1% MP shock (based on the monetary base) would reduce inflation by 0.38% after one year. Rossini and Vega (2007), considering various degrees of FX intervention, balance sheet effects, and competitiveness effects, analyzed the impact of MP shocks on GDP growth and inflation.

They found that the effect of the interest rate on inflation increased under high FX intervention, while under low FX intervention, the impact of the interest rate on GDP growth was amplified. Furthermore, they showed that the influence of MP shocks on inflation declined when the balance sheet effect outweighed the competitiveness effect.

Bigio and Salas (2006) made the first attempt to model the time-varying dynamics between the interest rate, the real exchange rate, the output gap, and inflation from 1994 to 2004, using a Smooth Transition Vector Autoregressive (STR-VAR) model. They noted that the effects of MP shocks on the output gap and inflation fluctuated over time, with these variables responding more significantly during recessions and booms, respectively. On the other hand, the first study to employ TVP-VAR-SV models, following Primiceri (2005), was conducted by Castillo et al. (2016). The researchers found that MP shocks introduced instability to GDP growth and inflation during the 1980s. Furthermore, under the IT regime, they observed the MP response to AS and AD shocks varied over time. Other studies, such as Portilla et al. (2022), following the methodology of Koop et al. (2009), demonstrated that: (i) an MP shock reduces GDP growth in the short term and inflation in the long term; (ii) the interest rate reacts more swiftly to AS shocks than to AD shocks, and (iii) during the pre-IT regime, MP shocks were a significant driver of GDP growth and inflation.

Our research diverges from the existing Peruvian literature in two respects. Firstly, much of the existing literature focuses on a single monetary variable. For instance, Castillo et al. (2016) utilized M1, while Portilla et al. (2022) focused on the interest rate. Our approach, however, incorporates multiple monetary variables such as the exchange rate, money, and the interest rate, which may enhance the identification of MP shocks. Moreover, many studies neglect the relationship between MP and exchange rate shocks, a crucial aspect in the formulation of the MP, as outlined by Arena and Tuesta (1999), Rossini et al. (2013, 2014), and Rodriguez et al. (2021). For example, Alvarado et al. (2023) consider a monetary aggregate and the interest rate but disregard the exchange rate. Secondly, our analysis of MP marks the first attempt to compare the fit of several TVP-VAR-SV models.

#### 3 Methodology

This Section describes the models and some details of the estimation algorithm, and presents the computation of the Log-ML and the DIC.

#### 3.1 The TVP-VAR-SV Model

The model in structural form is as follows:

$$\mathbf{B}_{0,t}\mathbf{y}_t = \boldsymbol{\mu}_t + \sum_{j=1}^p \mathbf{B}_{j,t}\mathbf{y}_{t-j} + \boldsymbol{\epsilon}_t, \quad \boldsymbol{\epsilon}_t \sim \mathcal{N}(\mathbf{0}, \boldsymbol{\Sigma}_t),$$
(1)

where  $\mathbf{y}_t$  is an  $n \times 1$  vector of endogenous variables,  $\boldsymbol{\mu}_t$  is an  $n \times 1$  vector of time-varying intercepts,  $\mathbf{B}_{j,t}$  is an  $n \times n$  matrix of time-varying coefficients associated with the *j*-th lag of the vector of endogenous variables,  $\mathbf{B}_{0,t}$  is the  $n \times n$  lower triangular matrix of contemporary coefficients with diagonal unit values,  $\boldsymbol{\Sigma}_t = diag(\exp(\mathbf{h}_{1t}), ..., \exp(\mathbf{h}_{nt}))$ , and  $\mathbf{h}_t = (\mathbf{h}_{1t}, ..., \mathbf{h}_{nt})'$ . The logs of all variables  $\mathbf{h}_t = (\mathbf{h}_{1,t}, ..., \mathbf{h}_{n,t})$  follow an independent random walk:

$$\mathbf{h}_t = \mathbf{h}_{t-1} + \boldsymbol{\zeta}_t, \quad \boldsymbol{\zeta}_t \sim \mathcal{N}(\mathbf{0}, \boldsymbol{\Sigma}_h), \tag{2}$$

where  $\Sigma_h = diag(\sigma_{h_1}^2, ..., \sigma_{h_n}^2)$ , and the initial conditions  $\mathbf{h}_0 \sim \mathcal{N}(\mathbf{a}_h, \mathbf{V}_h)$  must be estimated. Equation (1) can be rewritten as:

$$\mathbf{y}_t = \widetilde{\mathbf{X}}_t \boldsymbol{eta}_t + \mathbf{W}_t \boldsymbol{\gamma}_t + \boldsymbol{\epsilon}_t, \quad \boldsymbol{\epsilon}_t \sim \mathcal{N}(\mathbf{0}, \boldsymbol{\Sigma}_t),$$

where  $\widetilde{\mathbf{X}}_t = \mathbf{I}_n \otimes (1, \mathbf{y}'_{t-1}, \dots, \mathbf{y}'_{t-p}), \boldsymbol{\beta}_t = vec((\boldsymbol{\mu}_t, \mathbf{B}_{1,t}, \dots, \mathbf{B}_{p,t})')$  is a  $k_{\beta} \times 1$  vector of time-varying intercepts and coefficients associated with the lagged observations,  $\mathbf{W}_t$  is an  $n \times k_{\gamma}$  matrix that contains the appropriate elements of  $-\mathbf{y}_t^{-1}$  and  $\boldsymbol{\gamma}_t = (\boldsymbol{\gamma}_{1,t}, \boldsymbol{\gamma}_{2,t}, \boldsymbol{\gamma}_{3,t}, \dots, \boldsymbol{\gamma}_{k_{\gamma},t})'$  is a  $k_{\gamma} \times 1$  vector of time-varying coefficients that characterize contemporaneous relationships among variables. It is important to specify that  $k_{\beta} = n(np+1)$  and  $k_{\gamma} = n(n-1)/2$ . Considering  $\mathbf{X}_t = (\widetilde{\mathbf{X}}_t, \mathbf{W}_t)$ , the model can be simplified in the following state-space representation:

$$\mathbf{y}_t = \mathbf{X}_t \boldsymbol{\theta}_t + \boldsymbol{\epsilon}_t, \quad \boldsymbol{\epsilon}_t \sim \mathcal{N}(\mathbf{0}, \boldsymbol{\Sigma}_t), \tag{3}$$

$$\boldsymbol{\theta}_t = \boldsymbol{\theta}_{t-1} + \boldsymbol{\eta}_t, \quad \boldsymbol{\eta}_t \sim \mathcal{N}(\mathbf{0}, \boldsymbol{\Sigma}_{\theta}),$$

where  $\boldsymbol{\theta}_t = (\boldsymbol{\beta}'_t, \boldsymbol{\gamma}')'$  has a  $k_{\theta} = k_{\beta} + k_{\gamma}$  dimension, the initial conditions  $\boldsymbol{\theta}_0 \sim \mathcal{N}(\mathbf{a}_{\theta}, \mathbf{V}_{\theta})$  must be estimated, and  $\boldsymbol{\Sigma}_{\theta} = diag(\sigma^2_{\theta_1}, ..., \sigma^2_{\theta_k})$ . Moreover, we consider that the elements  $\boldsymbol{\Sigma}_{\theta}$  of  $\boldsymbol{\Sigma}_h$ are independently distributed as  $\sigma^2_{\theta_i} \sim \mathcal{IG}(\nu_{\theta_i}, S_{\theta_i})$  for  $i = 1, ..., k_{\theta}$  and  $\sigma^2_{h_j} \sim \mathcal{IG}(\nu_{h_j}, S_{h_j})$  for  $j = 1, ..., k_h$ .

#### 3.2 Competing Models

We estimate six additional models restricting different sets of parameters from the general TVP-VAR-SV model in (1): (i) a TVP-VAR model that considers homoscedastic variance ( $\mathbf{h}_t = \mathbf{h}_0$ ); (ii) a CVAR-SV model, where only the variances are time-varying; (iii) a TVP-VAR-R1-SV model that assumes constant parameters for the intercepts and the lagged variables ( $\boldsymbol{\beta}_t = \boldsymbol{\beta}_0$ ); (iv) a TVP-VAR-R2-SV model that considers constant coefficients for the contemporaneous relations ( $\boldsymbol{\gamma}_t = \boldsymbol{\gamma}_0$ ); (v) a TVP-VAR-R3-SV model that assumes that only the intercepts and variances are time-varying; and (vi) a CVAR model that keeps everything constant.

#### 3.3 Estimation Algorithm: Gibbs-Sampling<sup>2</sup>

We use the Gibbs sampling method in order to estimate the posterior parameters. This algorithm divides the parameters in blocks and estimates each one separately, conditional on updates in the other blocks. The draws are based on the precision sampling proposed by Chan and Jeliazkov (2009) and improved by Chan and Eisenstat (2018). The algorithm for estimating the TVP-VAR-SV model is as follows: (i) we get draws of  $\boldsymbol{\theta}$  from ( $\boldsymbol{\theta}|\mathbf{y}, \mathbf{h}, \boldsymbol{\Sigma}_{\theta}, \boldsymbol{\Sigma}_{h}, \boldsymbol{\theta}_{0}, \mathbf{h}_{0}$ ) ~  $\mathcal{N}\left(\hat{\boldsymbol{\theta}}, \mathbf{K}_{\theta}^{-1}\right)$ , where  $\hat{\boldsymbol{\theta}} = \mathbf{K}_{\theta}^{-1}\left(\mathbf{H}_{\theta}'\mathbf{S}_{\theta}^{-1}\mathbf{H}_{\theta}\boldsymbol{\alpha}_{\theta} + \mathbf{X}'\boldsymbol{\Sigma}^{-1}\mathbf{y}\right)$  and  $\mathbf{K}_{\theta} = \mathbf{H}_{\theta}'\mathbf{S}_{\theta}^{-1}\mathbf{H}_{\theta} + \mathbf{X}'\boldsymbol{\Sigma}^{-1}\mathbf{X}$  with  $\boldsymbol{\alpha}_{\theta} = \mathbf{H}_{\theta}^{-1}\tilde{\boldsymbol{\alpha}}_{\theta}$  (the matrices  $\mathbf{H}_{\theta}, \mathbf{X}, \mathbf{S}_{\theta}, \boldsymbol{\Sigma}$  and  $\tilde{\boldsymbol{\alpha}}_{\theta}$  are described in Appendix A of Chan and Eisenstat (2018)); (ii) for the draws of ( $\mathbf{h}|\mathbf{y}, \boldsymbol{\theta}, \boldsymbol{\Sigma}_{\theta}, \boldsymbol{\Sigma}_{h}, \boldsymbol{\theta}_{0}, \mathbf{h}_{0}$ ), the auxiliary mixture sampler of Kim et al. (1998) is used; (iii) we obtain draws of  $\sigma_{\theta_{i}}^{2}$  from  $\left(\sigma_{\theta_{i}}^{2}|\mathbf{y}, \boldsymbol{\theta}, \mathbf{h}, \boldsymbol{\theta}_{0}, \mathbf{h}_{0}\right) \sim \mathcal{IG}\left(\nu_{h_{i}} + \frac{T}{2}, S_{\theta_{i}} + \frac{1}{2}\sum_{t=1}^{T}(\theta_{i,t} - \theta_{i,t-1})^{2}\right)$  for  $i = 1, ..., k_{\theta}$ ; (iv) we obtain draws of  $\sigma_{h_{j}}^{2}$  from  $\left(\sigma_{h_{j}}^{2}|\mathbf{y}, \boldsymbol{\theta}, \mathbf{h}, \theta_{0}, \mathbf{h}_{0}\right) \sim \mathcal{IG}\left(\nu_{h_{j}} + \frac{T}{2}, S_{h_{j}} + \frac{1}{2}\sum_{t=1}^{T}(h_{j,t} - h_{j,t-1})^{2}\right)$ 

<sup>1</sup>For instance, when n = 3,  $\mathbf{W}_t = \begin{bmatrix} 0 & 0 & 0 \\ -y_{1,t} & 0 & 0 \\ 0 & -y_{1,t} & -y_{2,t} \end{bmatrix}$  where  $y_{i,t}$  is the *i*-th element of  $\mathbf{y}_t$  for i = 1, 2.

<sup>&</sup>lt;sup>2</sup>Complete details about the algorithm for estimating the TVP-VAR-SV model and the competing models are detailed in Section 4 and Appendix A of Chan and Eisenstat (2018).

for  $j = 1, ..., k_h$ ; (v) we sample draws of  $\boldsymbol{\theta}_0$  from  $(\boldsymbol{\theta}_0 | \mathbf{y}, \boldsymbol{\theta}, \mathbf{h}, \boldsymbol{\Sigma}_{\boldsymbol{\theta}}, \boldsymbol{\Sigma}_h) \sim \mathcal{N}(\widehat{\boldsymbol{\theta}}_0, \mathbf{K}_{\boldsymbol{\theta}_0}^{-1})$ , where  $\mathbf{K}_{\boldsymbol{\theta}_0} = \mathbf{V}_{\boldsymbol{\theta}}^{-1} + \boldsymbol{\Sigma}_{\boldsymbol{\theta}}^{-1}$  and  $\widehat{\boldsymbol{\theta}}_0 = \mathbf{K}_{\boldsymbol{\theta}_0}^{-1} (\mathbf{V}_{\boldsymbol{\theta}}^{-1} \mathbf{a}_{\boldsymbol{\theta}} + \boldsymbol{\Sigma}_{\boldsymbol{\theta}}^{-1} \boldsymbol{\theta}_1)$ ; (vi) we obtain draws of  $\mathbf{h}_0$  from  $(\mathbf{h}_0 | \mathbf{y}, \boldsymbol{\theta}, \mathbf{h}, \boldsymbol{\Sigma}_{\boldsymbol{\theta}}, \boldsymbol{\Sigma}_h) \sim \mathcal{N}(\widehat{\mathbf{h}}_0, \mathbf{K}_{\boldsymbol{h}_0}^{-1})$ , where  $\mathbf{K}_{\mathbf{h}_0} = \mathbf{V}_{\mathbf{h}}^{-1} + \boldsymbol{\Sigma}_{\mathbf{h}}^{-1}$  and  $\widehat{\mathbf{h}}_0 = \mathbf{K}_{\mathbf{h}_0}^{-1} (\mathbf{V}_{\mathbf{h}}^{-1} \mathbf{a}_h + \boldsymbol{\Sigma}_{\mathbf{h}}^{-1} \mathbf{h}_1)$ ; and (vii) steps (i)-(vi) are repeated N times. Moreover,  $\mathbf{a}_{\boldsymbol{\theta}}, \mathbf{V}_{\boldsymbol{\theta}}, \mathbf{a}_h, \mathbf{V}_{\mathbf{h}}, \nu_{\boldsymbol{\theta}_i}, \nu_{\boldsymbol{h}_j}, S_{\boldsymbol{\theta}_i}$  and  $S_{h_j}$  are defined in Section 4.2.

#### 3.4 Bayesian Model Comparison

This Section describes the Log-ML and the DIC used to select the best model.

#### 3.4.1 Log Marginal Likelihood $(Log-ML)^3$

Based on the importance sampling density  $g^* = g(\theta_n)$  and the cross-entrophy method, Chan and Eisenstat (2015) propose the following estimator for the marginal likelihood:

$$\widehat{p}_{IS}(\mathbf{y}) = \frac{1}{N} \sum_{n=1}^{N} \frac{p(\mathbf{y}|\boldsymbol{\theta}_n)p(\boldsymbol{\theta}_n)}{g^*},\tag{4}$$

where  $\theta_1, \ldots, \theta_N$  are the independent draws of  $g^*$ . On one hand, regardless the value of  $g^*$ , the estimator  $\hat{p}_{IS}(\mathbf{y})$  is consistent and unbiased, however, its variance depends on  $g^*$ . On the other hand, using the posterior density for  $g^*$ ,  $\hat{p}_{IS}(\mathbf{y})$  is equivalent to  $p(\mathbf{y})$ , but  $g^* = p(\boldsymbol{\theta}|\mathbf{y})$  causes endogenity since it depends on  $p(\mathbf{y})$ . Therefore, using the cross-entrophy method, we must select g near to  $g^*$  so that the variance of  $\hat{p}_{IS}(\mathbf{y})$  is minimized.

Considering the parametric family  $F = \{f(\boldsymbol{\theta}; \mathbf{v})\}$  indexed by  $\mathbf{v}$ , we calculate the importance sampling  $f(\boldsymbol{\theta}; \mathbf{v}) \in F$  near to  $g^*$  using the density  $f(\boldsymbol{\theta}; \mathbf{v}_{ce}^*) \in F$  that cuts down the cross-entrophy distance between  $g^*$  and the chosen density  $f(\boldsymbol{\theta}; \mathbf{v})$  as follows:

$$\mathbf{v}_{ce}^* = \arg\min\left(\int g^*(\boldsymbol{\theta})\log g^*(\boldsymbol{\theta})\mathbf{d}\boldsymbol{\theta} - p(\mathbf{y})^{-1}\int p(\mathbf{y}|\boldsymbol{\theta})p(\boldsymbol{\theta})\log f(\boldsymbol{\theta};\mathbf{v})\mathbf{d}\boldsymbol{\theta}\right),\tag{5}$$

that can be simplified to:

$$\hat{\mathbf{v}}_{ce}^* = \arg\max_{\mathbf{v}} \frac{1}{L} \sum_{l=1}^{L} \log f(\boldsymbol{\theta}_l; \mathbf{v}), \tag{6}$$

where  $\boldsymbol{\theta}_1, \ldots, \boldsymbol{\theta}_L$  are the posterior draws.

#### **3.4.2** Deviance Information Criterion (DIC)

The Deviance Information Criterion (DIC), proposed by Spiegelhalter et al. (2002), ponderates both the fit and the complexity of a model. On one hand, we consider the deviance  $(D(\theta))$  as a measure of the goodness of fit of the model, which is defined as Chan and Grant (2016) as follows:

$$D(\boldsymbol{\theta}) = -2\log f(\mathbf{y}|\boldsymbol{\theta}) + 2\log h(\mathbf{y}),\tag{7}$$

where  $f(\mathbf{y}|\boldsymbol{\theta})$  is the likelihood function of the model and  $h(\mathbf{y})$  is a function of the data. Moreover, we consider the effective number of parameters $(p_D)$  as a measure of the complexity of the model:

$$p_D = \overline{D(\boldsymbol{\theta})} - D(\widetilde{\boldsymbol{\theta}}),$$

<sup>&</sup>lt;sup>3</sup>Complete details are described in Section 4 and Appendix B of Chan and Eisenstat (2018).

where  $\overline{D(\theta)} = -2\mathbf{E}_{\theta}[\log f(\mathbf{y}|\theta)|\mathbf{y}] + 2\log h(\mathbf{y})$  is the posterior mean deviance and  $\tilde{\theta}$  is an estimate of  $\theta$ . Therefore, the DIC is the sum of the posterior mean deviance and the effective number of parameters:  $DIC = \overline{D(\theta)} + p_D$ . Considering  $h(\mathbf{y}) = 1$ , we obtain:

$$DIC = -4E_{\boldsymbol{\theta}}[\log f(\mathbf{y}|\boldsymbol{\theta})] + 2\log f(\mathbf{y}|\widetilde{\boldsymbol{\theta}}), \tag{8}$$

where  $-4E_{\theta}[\log f(\mathbf{y}|\boldsymbol{\theta})]$  contains the average of the log of  $f(\mathbf{y}|\boldsymbol{\theta})$  over the posterior draws of  $\boldsymbol{\theta}$ .

#### 4 Empirical Evidence

This Section first introduces the data, moves onto the priors, and finally delves into the empirical results. The analysis encompasses the evolution of the estimated parameters, the volatility of exogenous shocks, the IRFs of MP shocks, the FEVDs, and the variables' HD.

#### 4.1 Data

We use six variables for the period 1996Q1-2018Q2: S&P GSCI growth (Figure 1a), nominal exchange rate growth (Figure 1b), real GDP growth (Figure 1c), inflation rate (Figure 1d), money growth<sup>4</sup> (Figure 1e), and the interest rate (Figure 1f). The data is sourced from the BCRP website, except for the S&P GSCI, which was obtained from Bloomberg. All variables are expressed as yearon-year percent changes, except for the interest rate. The latter is a combination of the interbank interest rate (until 2003Q3) and the reference interest rate (2003Q4-2018Q2). Until 2009, S&P GSCI growth averages 20%; during the GFC it falls to a low of -43%; and after 2010, it averages almost 1%. The behavior of this variable aligns with EMEs' higher and lower growth before and after the GFC, respectively. Exchange rate growth exhibits a negative trend before the GFC and a positive trend after 2010. This variable increases during both the Asian-Russian crisis and the GFC but reaches a peak during the former. Inflation and GDP growth are less volatile under the IT regime. Specifically, before 2002, inflation is in the double digits; after 2002, it ranges between 1% and 3%. Regarding money growth, it displays a positive trend until 2009, but later a negative trend, stabilizing at the end of the sample. Prior to the IT regime, the interest rate averages double digits; however, under the IT regime, it averages around 3%. The high digits before the IT regime are initially attributable to the financial instability that occurred during the Asian-Russian crisis but persisted for several years.

#### 4.2 Priors

In the baseline TVP-VAR-SV model, we set  $\mathbf{a}_{\theta} = 0$ ,  $\mathbf{V}_{\theta} = 10 \times \mathbf{I}_{k_{\theta}}$ ,  $\mathbf{a}_{h} = 0$ ,  $\mathbf{V}_{h} = 10 \times \mathbf{I}_{n}$  and  $\nu_{\theta_{i}} = \nu_{h_{j}} = 5$ . Moreover, we fix  $S_{\theta_{i}}$  and  $S_{h_{j}}$  so that the initial means of  $\sigma_{\theta_{i}}^{2}$  and  $\sigma_{h_{j}}^{2}$  are 0.01<sup>2</sup> and 0.1<sup>2</sup>, respectively.

#### 4.3 Empirical Results

For the estimations, p = 2 is selected based on the BIC for a CVAR model. We perform 11000 draws for all models in 10 parallel chains, discarding the initial 1000. This leaves 100000 draws, from which we select one out of every 10, resulting in a total of 10000 draws used to estimate the DIC and the Log-ML.

 $<sup>^{4}</sup>$ We use BCRP-issued banknotes held by the private sector as a proxy for the money supply, as proposed by Quispe (1998) and León Fernández (1999).

Structural shocks are identified as follows: an external shock for the S&P GSCI growth equation; an ER shock for the nominal exchange rate growth equation; an AD shock for the GDP growth equation; an AS shock for the inflation equation; an MS shock for the money growth equation; and an MP shock for the interest rate equation<sup>5</sup>.

#### 4.3.1 Evidence on Parameter Evolution

To analyze potential changes in parameters and volatilities over time, we employ the trace test proposed by Cogley and Sargent (2005), the Kolmogorov-Smirnov test (KS-test), and the t-test. The trace test assesses whether the prior trace of  $\Sigma_h$  differs statistically from the median posterior trace of  $\Sigma_h$ . The KS-test analyzes whether, at two different times, each parameter and volatility are drawn from the same continuous distribution, while the t-test evaluates whether, at two different times, the mean of each parameter and volatility originates from the same Normal distribution. We consider two comparison groups for the latter two tests: (i) 1996Q3-2002Q1 vs 2002Q2-2018Q2; and (ii) 1996Q3-2007Q2 vs 2007Q3-2018Q2.

Table 1 displays the results of these three tests for  $\gamma_t$ ,  $\beta_t$ , and  $\mathbf{h}_t$  in the TVP-VAR-SV model. Firstly, all elements of  $\mathbf{h}_t$  change over time. The trace test shows that the trace of the posterior median of  $\Sigma_h$  is statistically significant but small. Specifically, at 84% confidence, this trace is between 0.17% and 0.34%. Both the KS-test and the t-test show that all elements of  $\mathbf{h}_t$  vary over time for both comparison groups. Secondly, under the KS-test and the t-test, all elements of  $\gamma_t$ change over time in both comparison groups. Thirdly, most elements of  $\beta_t$  change, although more conclusively for the second comparison group. According to the KS-test, 72% of the components of  $\beta_t$  change for the first group, while 81% of the elements of  $\beta_t$  vary for the second group. In sum, these results suggest that all volatilities and most of the parameters change over time.

#### 4.3.2 Model Comparison

Table 2 shows both the mean and the standard deviation of the Log-ML and the DIC for all models detailed in Sections 3.1 and 3.2. According to the Log-ML, we select the TVP-VAR-R1-SV and TVP-VAR-R3-SV models as their results are similar, whereas according to the DIC, the CVAR-SV model is selected. Specifically, the Bayes Factor (BF) of the TVP-VAR-R1-SV model against the TVP-VAR-R3-SV model is 2.85, reinforcing the results obtained from the KS-test and t-test that all elements of  $\gamma_t$  might vary over time. Both criteria indicate that SV improves fit as all five SV models are in the top five spots. For instance, the BF of the TVP-VAR-SV model is  $1.06 \times 10^{32}$ , demonstrating that including SV considerably enhances fit. Moreover, fit is reduced by assuming time-varying parameters for the lagged variables; e.g., the BF of the TVP-VAR-R3-SV model against the TVP-VAR-R2-SV model is  $5.24 \times 10^6$ . In addition, with respect to both criteria, the CVAR model is in the last spot, which shows that assuming all parameters and variances are constant sharply reduces the fit; e.g., the BF of the TVP-VAR-SV model against the CVAR model is  $1.09 \times 10^{34}$ .

### 4.3.3 Volatility of Exogenous Shocks

Figure 2 illustrates the posterior median of the volatilities of all shocks for all models. Mainly, SV models show that all volatilities change over time following similar patterns. Firstly, the volatilities

<sup>&</sup>lt;sup>5</sup>Under the pre-IT regime, an MP shock was associated with the monetary base (1995-2001) and the banks' current accounts with the BCRP (2002-2003). However, for the purposes of this study, we assume an MP shock is associated with the interest rate for the entire sample period, as argued by Portilla et al. (2022).

of AD, AS, MS, and MP shocks decline after IT adoption in 2002. Explicitly, during the pre-IT regime, the average volatility of MP shocks was 4%, while during the IT regime, this average was almost 0.3%. Similarly, the volatility of AD shocks drops from 1.9% on average before IT adoption to 1.2% during the IT regime. Conversely, the volatilities of external and ER shocks increase until the GFC, after which both decrease. Therefore, these results could suggest that even though volatility in the commodity market increased in the 2000s, IT adoption reduced the volatility of domestic structural shocks and, consequently, macroeconomic volatility in Peru. These findings align with Castillo et al. (2009), who argue that the adoption of the reference interest rate as the operative instrument reduced risk, and with Castillo et al. (2016), who estimate that the volatilities of AD and AS shocks fell since the 1990s.

Secondly, the volatilities of AD and MP shocks reach their maximum during the Asian-Russian crisis, while the volatilities of external, ER, and MP shocks peaked during the GFC. In these two crises, Peru experienced sudden stops and severe contractions in external variables (e.g., the S&P GSCI declined by 25% and 45% during the Asian-Russian crisis and the GFC, respectively). The volatility of MP shocks increases during both crises, but less so during the GFC, corresponding to different ex-ante and ex-post policies, such as the buildup of net international reserves (NIR), according to Velarde (2015), Rossini (2016), and Castillo and Barco (2009). In particular, the NIR/GDP ratio was 17% at the onset of the Asian-Russian crisis, so the BCRP could not provide adequate FX liquidity to mitigate the exchange rate depreciation and had to raise the interbank interest rate by almost 20 p.p. Conversely, during the GFC, the NIR/GDP ratio was 24%, so the BCRP could inject a sufficient FX amount and had to adjust the reference interest rate by just around 5 p.p.

Thirdly, external and ER shocks follow similar patterns that differ from the rest of the shocks. Among all shocks, the volatility of external shocks is the highest. Between 2002 and 2011, this volatility shows a positive trend explained by the commodity price boom, which consisted of an increasing growth rate of commodity prices and, consequently, in EMEs' activity (see Mendoza (2013) for further details for Peru). In addition, this volatility captures the uncertainty caused by the GFC, reaching its maximum of 19% on average across SV models<sup>6</sup>. Later, from 2011 onwards, this volatility displays a negative trend due to the end of the commodity supercycle. Meanwhile, leading up to the GFC, the volatility of ER shocks had been escalating due to heightened terms of trade, growing capital inflows, and the entry of non-residents into the Peruvian financial market in 2005, which increased the number of participants in the FX market and its volatility. After the GFC, this volatility follows a negative trend, which might be explained by an improvement in the BCRP's identification of pressures in the FX market and, consequently, its intervention in this market. For example, in periods of uncertainty, non-resident investors sought to hedge by purchasing dollars in the spot market, which increased the volatility of the exchange rate. Therefore, in 2014 the BCRP incorporated exchange rate swaps as an alternative hedging instrument.

The constant volatilities of the CVAR and TVP-VAR models respectively underestimate and overestimate those predicted by the SV models. These models fail to capture the effects of crises. For instance, during the GFC, the CVAR model estimates a constant volatility of 15%, nearly 6 p.p. lower than that estimated by SV models. Additionally, they do not accurately estimate the effects under the IT regime; e.g., the TVP-VAR model posits that the volatility of the interest rate was 1.8% in the early 2000s, thereby overlooking the gains of anchoring expectations.

From this point forward, we focus on the results of four models: TVP-VAR-R3-SV, TVP-VAR-R1-SV, CVAR-SV, and CVAR, the latter included for comparison purposes.

<sup>&</sup>lt;sup>6</sup>Excluding this crisis, the average volatility for SV models is 14%.

#### 4.3.4 Impulse Response Functions (IRFs)

This Section discusses the IRFs to MP shocks on GDP growth and inflation, along with the IRFs to external, ER, AD, and AS shocks on the interest rate. The IRFs are normalized to unity for the entire sample.

Figure 3 provides a 3D plot presenting the posterior medians of the IRFs of GDP growth and inflation for the four selected models. Both GDP growth and inflation decrease in response to a contractionary MP shock. GDP growth reacts more rapidly, reaching maximum contraction after four quarters, while inflation falls after three quarters and attains its minimum level after six quarters. These results align with those obtained by Rossini and Vega (2007), who estimate that GDP growth reaches its maximum contraction after four quarters, and Quispe (2000), who demonstrates that inflation hits its lowest level after five quarters. On average, all models estimate that GDP growth falls by a maximum of 0.28 p.p., which contrasts with Pérez Forero's (2015) estimate of 0.15 p.p., while inflation falls by a maximum of 0.10 p.p. The differential responses might be due to the presence of sticky prices, which reduce the impact of MP shocks on prices.

Moreover, while the IRFs of the TVP-VAR-R3-SV and CVAR-SV models are time-invariant, the TVP-VAR-R1-SV model reveals that the effect of MP shocks on GDP growth and inflation can change over time. On one hand, GDP growth reacts more robustly during recessions (the Asian-Russian crisis and the GFC), suggesting the presence of a convex supply curve, as explained by Bigio and Salas (2006). Conversely, inflation responds more vigorously when situated at the upper limit of the inflation target band. For instance, in 2006, when the average inflation was 2%, inflation decreased by a maximum of 0.02 p.p.; in 2008, when the average inflation was 2.75%, inflation declined by a maximum of 0.13 p.p.

Figure 4 presents the posterior medians of the sample and the 68% confidence bands of the IRFs to an MP shock on GDP growth and inflation. All models estimate that GDP growth declines for five quarters following the initial shock, potentially due to the year-long pass-through effect of the reference interest rate into market interest rates, as argued by Lahura (2006, 2017). On average, only the TVP-VAR-R1-SV model calculates that inflation diminishes, which might be due to this model presenting the lowest short-run price puzzle among all models, occurring only after four quarters.

Figure 5 depicts the posterior medians of the IRFs to an MP shock on GDP growth and inflation in the four previously selected models at four time periods<sup>7</sup>: 1998Q3, 2003Q4, 2009Q3, and 2018Q2. All models estimate that GDP growth and inflation decrease, with the TVP-VAR-R1-SV model showing both variables diminishing more under the IT regime. This could be because the signaling of MP improved under the IT regime, as explained by Rossini and Vega (2007), and that the pass-through effect of the reference interest rate into market interest rates increased. For example, during the GFC, inflation decreased by a maximum of 0.12 p.p.; however, during the Asian-Russian crisis, inflation fell by a maximum of 0.08 p.p.

Figure 6 plots the interest rate's response to external, ER, AD, and AS shocks at four different points in time: 1998Q3, 2003Q4, 2009Q3, and 2018Q2. In all models, the interest rate rises in response to AS shocks, peaking after two quarters. Additionally, SV models estimate that the interest rate escalates in response to AD shocks, presenting its highest response after four quarters. Specifically, reactions to AS shocks are stronger than responses to AD shocks, as in Portilla et al. (2022). This provides evidence of the BCRP's anti-inflationary bias, which involves immediately mitigating the direct effects on inflation to anchor inflation expectations. Furthermore, this bias

<sup>&</sup>lt;sup>7</sup>1998Q3 and 2009Q3 are the quarters of highest volatility during the Asian-Russian crisis and the GFC, respectively; while 2003Q4 and 2018Q2 represent the introduction of the reference interest rate and the end of the sample, respectively.

has intensified over time, as the interest rate's response to AS shocks strengthens. For instance, a 1% AS shock increases the interest rate by a maximum of 0.4 p.p. in 2009Q3, whereas, in 2018Q2, the interest rate rises by a maximum of 1.2 p.p.

Regarding external shocks, the interest rate's IRF varies between models and over time. Dancourt (2013) and Quispe et al. (2017) suggest that the interest rate rises because this shock boosts GDP growth and, consequently, inflation. However, only the TVP-VAR-R1-SV model supports this outcome. Conversely, TVP-VAR-R3-SV and CVAR models posit that the interest rate falls to mitigate the capital inflows that continuously appreciate the exchange rate and cause deflation. These two models also estimate that the reaction to external shocks increases over time, demonstrating a stronger counter-cyclical response of the BCRP to foreign factors.

Concerning ER shocks, all models indicate that the interest rate rises. This result aligns with the BCRP's mandate, which involves reducing excessive exchange rate volatility that can negatively affect economic agents' balance sheets, as explained by Castillo et al. (2011). Additionally, the interest rate's response to ER shocks intensifies over time. For instance, a 1% ER shock increases the interest rate by an average of 0.2 p.p. on average in 2009Q3, compared with 1 p.p. on average in 2018Q2. These results contradict Rodriguez et al. (2023a), who find that the interest rate does not respond to variations in the exchange rate. However, this discrepancy might be explained by the omission of FX reserves in our model. Despite this, we also find that during the IT regime, the reaction to fluctuations in the exchange rate increases.

#### 4.3.5 Forecast Error Variance Decomposition (FEVD)

Figure 7 illustrates the FEVD of GDP growth, inflation, and the interest rate over two different horizons: two (h = 2) and twenty (h = 20) quarters.

Regarding the FEVD of GDP growth, AD shocks serve as the primary source of uncertainty until 2003, when external shocks take over as the main determinant, increasing their dominance over longer horizons. This outcome aligns with Mendoza (2013), Rodriguez et al. (2018) and Rodríguez et al. (2023b), who argue that increased trade openness since 2003 and the commodity supercycle account for the significant share of external shocks in the FEVD of GDP growth. These shocks explain 50% of the FEVD of GDP growth on average. MP shocks are only relevant during the pre-IT regime, which aligns with the arguments of Rossini and Vega (2007) and Portilla et al. (2022) that these shocks became more predictable after the adoption of the interest rate as the operative instrument. The MP shocks peaked during the Asian-Russian crisis, when the BCRP unexpectedly raised the policy rate to prevent a sudden stop, as detailed by Velarde and Rodriguez (2001). In contrast, MP shocks made a minor contribution during the GFC due to the BCRP providing more liquidity to banks and intervening in the FX market. Moreover, in the long run, the share of most shocks increases, as calculated by Bringas and Tuesta (1997).

In terms of the FEVD of inflation, external and AS shocks are the most influential. Specifically, they account for approximately 85% of the FEVD on average, given that external shocks are the main source of uncertainty after 2003. Contrary to what is observed in the FEVD of GDP growth, MP shocks present a smaller share in the FEVD of inflation, which could be related to the existence of sticky prices in Peru. Furthermore, ER shocks contribute an average of 12%, which does not provide sufficient evidence of an exchange rate pass-through effect on prices in Peru, as explained by Winkelried (2003, 2012). The MS shocks become relevant over longer horizons, with their share increasing from 4% to 20% in the TPV-VAR-R3-SV model, which might indicate the existence of monetary inflation in Peru. Over longer horizons, AS and MP shocks present lower and higher shares, respectively.

In the case of the FEVD of the interest rate, MP shocks are the most significant during the

pre-IT regime, as they explain almost 85% of the FEVD. However, this percentage surges to 95% during the Asian-Russian crisis because of the increase in the interbank interest rate by 19.7 p.p. in 1998. In the IT regime, external and ER shocks are the primary determinants of the FEVD of the interest rate. On one hand, external shocks have the highest share, which could suggest the high impact of these shocks on a small open commodity-exporting economy like Peru. On the other hand, the relevant share of ER shocks is due to the BCRP's intensive response to mitigate its adverse effect on the money market, as detailed by Castillo et al. (2011). During the GFC, the share of MP shocks rises, but it is lower than what was observed during the Asian-Russian crisis, since the BCRP reacted more efficiently by injecting greater liquidity via reverse repos, as detailed by Dancourt and Jiménez (2010). Over longer horizons, the participation of ER, AS, and MS shocks increases, while the participation of MP shocks decreases.

#### 4.3.6 Historical Decomposition (HD)

Figure 8 shows the HD for GDP growth, inflation, and interest rate. The HD of GDP growth shows that between the pre-IT regime (1996-2001) and the commodity price boom (2002-2011), this variable increased from 2.3% to 6.2%. All models estimate that this 3.9-p.p. increase was primarily determined by MP and ER shocks. For instance, the CVAR-SV model calculates that MP shocks explained 1.5 p.p. (around 39% of the increase), whereas ER shocks contributed 1 p.p. (around 25% of the increase).

Between the commodity price boom (2002-2011) and the following years (2012-2018), the HD of GDP growth indicates that this variable decreased by 2.2 p.p. and most models consider that ER shocks are the main cause. For example, the TVP-VAR-R1-SV model shows that these shocks explained -0.7 p.p. (almost 34% of the decrease), while external and MS shocks contributed -0.55 p.p. (around 28% of the decrease) and -0.51 p.p. (around 26% of the decrease), respectively.

Moreover, we can quantify the effect of international crises on GDP growth. First, during the GFC, most models calculate that this variable fell from 9.2% to 1.1% and that this decrease was primarily determined by AD and MS shocks, explaining falls of -2.9 p.p. and -3 p.p., respectively During the GFC, MP shocks explained 0.1 p.p. (around -1.4% of the decrease), showing the small impact of the monetary authority. Second, during the Asian-Russian crisis, GDP growth declined from 6.4% to -0.3%, which was caused by a decrease in AD and MP shocks, explaining reductions of -7 p.p. and -1 p.p., respectively. Unlike during the GFC, MP exerted considerable influence on the dynamics of GDP growth during the Asian-Russian crisis.

In comparing the HD of inflation before and during the IT regime, most models indicate that inflation fell from 5.5% to 2.8%, with AS and MS shocks as the primary determinants, each explaining a decrease of 0.5 p.p. Regarding MP shocks, all models show that these shocks contributed 0.3 p.p. (around -10% of the decrease). In addition, external and ER shocks explained -0.2 p.p. and -0.3 p.p. of the fall in inflation, respectively. During the GFC, inflation falls from 6% to 2.3% and all models estimate this was primarily explained by AS and MS shocks. For instance, the TVP-VAR-R3-SV model estimates that AS and MS shocks explained -2.2 p.p. (59% of the decrease) and -1.1. p.p. (29% of the decrease), respectively. Moreover, all models show that MP shocks had a small positive impact on the fall in inflation during the GFC (e.g., the CVAR-SV model estimates that MP shocks explain 0.1 p.p.).

The HD of the interest rate indicates that under the IT regime, this variable decreased from 13.6% to 3.7%. The selected SV models estimate that this 9.8-p.p. decrease was mainly explained by MP shocks, e.g., the TVP-VAR-R1-SV model estimates that these explained -4.9 p.p. (around 59% of the decrease). Moreover, the selected SV models consider MS shocks as the second factor behind the interest rate fall. For instance, the TVP-VAR-R1-SV model calculates that MS shocks

account for -1.2 p.p. (14% of the decrease), which might be a consequence of using the monetary base as the intermediate policy objective. Finally, external shocks have a minor share across models, as they explain -0.3 p.p. (3.4% of the decrease) at most.

In terms of the effects of crises on the interest rate, the findings are as follows. During the Asian-Russian crisis, the interest rate increased from 12.8% to 18.8%, primarily due to a rise in MS and MP shocks of 1 p.p. (around 15% of the increase) and 8 p.p. (around 130% of the increase), respectively. During the GFC, the interest rate fell from 5.9% to 3.2%. According to the selected SV models, this reduction was explained by a decrease in AD, AS, MS, and MP shocks; e.g., the TVP-VAR-R3-SV model quantifies that these four shocks contribute to a reduction of -2.5 p.p.

Among all HDs, the effects of external shocks are less significant than those obtained in Ojeda Cunya and Rodriguez (2022), Portilla et al. (2022), Alvarado et al. (2023), Chávez and Rodríguez (2023) and Rodríguez et al. (2023b). For example, Chávez and Rodriguez. (2022) calculate that external shocks explain around 89% of the increase of GDP growth under the post-IT regime. However, all these papers omit ER shocks, thereby ignoring a rapid-impact mechanism of external shocks on the economy.

#### 5 Robustness Analysis

This Section discusses four robustness checks: (i) changes in the hyperparameters; (ii) changes in the recursive order (initially, inflation and money growth switch places, followed by an exchange between inflation and GDP growth); (iii) use of non-primary GDP growth instead of total GDP growth; and (iv) use of export price index (EPI) growth instead of S&P GSCI growth. Tables 3 and 4 present the results of this Section, while the Figures are available in an Appendix upon request.

In terms of the hyperparameter changes, all outcomes are similar to the base model. The trace test, KS-test, and t-test, reported in Table 3, indicate that all elements of  $\gamma_t$  and  $\mathbf{h}_t$  vary over time, with about 60% of the elements of  $\beta_t$  also changing. Furthermore, according to Table 4, the baseline ranking stands. The BF of the TVP-VAR-SV model against the CVAR model is  $2.68 \times 10^{16}$ , demonstrating an improvement in fit when considering both TVP and SV. Additionally, the volatilities of all exogenous shocks are qualitatively similar. Also, the IRFs to an MP shock on GDP growth and inflation are robust, as GDP growth reacts faster, reaching its maximum contraction after four quarters, whereas inflation falls after three quarters, reaching its minimum level after six quarters. Furthermore, the IRFs to AS, AD, ER, and external shocks on the interest rate are robust. Finally, the FEVDs and HDs are alike, as external shocks are the main driver of all variable uncertainties across the sample, while MP shocks are relevant for both uncertainty and fluctuations under the pre-IT regime.

Regarding changes in the recursive order, the results also remain robust. Firstly, Table 3 shows that the trace test, the KS-test, and the t-test present results similar to those estimated in the baseline model. In both recursive orders, all elements of  $\gamma_t$  and  $\mathbf{h}_t$  vary over time, while 59% of the elements of  $\beta_t$  also vary. Secondly, Table 4 illustrates that SV models fill the top five spots. For example, in the first alternative order, the BF of the TVP-VAR-R1-SV model against the TVP-VAR-R3-SV model is 3.01. Furthermore, in the first alternative order, the BF of the TVP-VAR-R3-SV model against the CVAR model is  $1.52 \times 10^{34}$ , compared with  $8.70 \times 10^{33}$  in the second alternative order, suggesting an improvement in fit when considering TVP and SV. Thirdly, all shock volatilities are identical to those in the baseline model. The volatilities of external shocks remain the highest and rise over time, like ER shocks; whereas AD, AS, and MS shocks continuously shrink, and MP shocks experience a sharp drop after IT adoption. Fourthly, the IRFs for GDP growth, inflation, and the interest rate are robust. Lastly, the FEVDs and HDs remain identical.

The results of the third exercise largely mirror those computed in the base model. Our analysis

reveals that all elements of  $\gamma_t$  and  $\mathbf{h}_t$  vary over time, while 60% of the elements of  $\beta_t$  undergo changes. Table 4, however, presents a few contrasting results. Although the ranking remains consistent according to the DIC, the Log-ML selection favors the TVP-VAR-R3-SV model. As with the previous two exercises, considering TVP and SV enhances fit, as the BF of the TVP-VAR-SV model against the CVAR model is  $1.06 \times 10^{39}$ . The volatility of all shocks remains the same except for those associated with AD shocks, which start from lower values. The IRFs to an MP shock on GDP growth and inflation are qualitatively similar. Nonetheless, the response of the interest rate to AD shocks is higher, possibly explained by the greater financial integration of the non-primary sector, enabling a more substantial pass-through effect of the reference interest rate into market interest rates. Finally, both the FEVDs and HDs estimate a more significant share for AD shocks compared to the base model.

To adequately capture the external factors impacting the Peruvian economy, the fourth exercise employs EPI growth as the external variable, yielding results akin to those in the base model. Firstly, Table 3 indicates that the results from the trace test, the KS-test, and the t-test are robust, with all elements of  $\gamma_t$  and  $\mathbf{h}_t$  varying over time, while nearly 64% of the elements of  $\beta_t$  change. Secondly, according to the DIC, the model ranking remains consistent, although the Log-ML selects the TVP-VAR-R3-SV model. Consequently, the BF of the TVP-VAR-R1-SV model against the TVP-VAR-R3-SV model stands at 0.54, while the BF of the TVP-VAR-SV model against the CVAR model is  $1.39 \times 10^{39}$ . Thirdly, volatilities are identical for all shocks except those associated with external shocks, which are lower than in the baseline case, but also increase during the Asian-Russian crisis and the GFC. Fourthly, the IRFs to an MP shock on GDP growth and inflation are robust; however, they show that MP shocks have a more extended but lesser impact on inflation. Fifthly, the IRFs to AS, AD, and ER shocks on the interest rate are robust, while the interest rate rises in response to positive external shocks. Lastly, the predominance of external shocks in the FEVDs and HDs of domestic variables diminishes, while the share of ER shocks increases.

#### 6 Conclusions

This paper examines the time-varying impact of MP shocks in Peru during the period from 1996Q1 to 2018Q2. For this purpose, we estimate a set of time-varying parameter vector autoregressive models with stochastic volatility (TVP-VAR-SV models) as proposed by Chan and Eisenstat (2018). We used six recursively identified variables: S&P GSCI growth, exchange rate growth, GDP growth, inflation, money growth, and the interest rate.

The key findings can be summarized as follows: all volatilities, all contemporary coefficients, and approximately 75% of the parameters for lagged variables and intercepts should change over time, as indicated by the trace test, KS test, and t-test. Furthermore, SV improves model fit, according to the Log-ML and the DIC. The volatilities of AD, AS, MS, and MP shocks decrease under the IT regime. On the other hand, the volatilities of external and ER shocks rise after the GFC, suggesting that IT adoption reduced the volatility of domestic structural shocks. Particularly, the volatility of MP shocks increased more during the Asian-Russian crisis than during the GFC, due to the different policies implemented by the BCRP.

The selected models estimate that a countercyclical MP shock decreases both GDP growth and inflation, but GDP growth responds quicker and more intensively than inflation, which could be attributed to the existence of sticky prices. Moreover, the reactions of the interest rate to AS shocks are stronger than the responses to AD shocks, aligning with the BCRP's anti-inflationary bias. All models also estimate an increasing reaction of the interest rate to external and ER shocks. According to the FEVDs and HDs of GDP growth and inflation, MP shocks are only significant during the pre-IT regime, suggesting an improvement in monetary policy since IT adoption. These results collectively indicate that IT adoption contributed to stabilizing macroeconomic uncertainty in Peru.

In terms of robustness checks, we find that alternative specifications, including changes in priors, recursive order, or certain variables, reinforce the main findings. For instance, the evidence of changing variances and parameters is reaffirmed. The effects of MP shocks on GDP growth and inflation remain qualitatively similar, and the interest rate's reaction to various shocks is generally robust across different exercises.

Future research could consider alternatives such as non-recursive identifications, as suggested by Quispe (2001), or sign restrictions as detailed in Guevara and Rodríguez (2020). Including additional variables could enhance the identification of shocks and MP transmission channels. For instance, credit variables like loan supply could enhance credit channel visibility, as shown in Viladegut and Cabello (2014) and Guevara and Rodríguez (2020). Furthermore, the identification of MP shocks could be improved by incorporating other BCRP instruments, as proposed by Lahura (2010), Castillo et al. (2011), and Rodriguez et al. (2021). However, as the dimensionality of the TVP-VAR-SV model would increase, future research might consider large Bayesian VARs as suggested by Chan (2022) and Chan and Yu (2022).

#### References

- Alvarado, P., Cáceres, M., and Rodríguez, G. (2023). Time-Varying Effects of Monetary Shocks on Macroeconomic Fluctuations in Peru: An Empirical Application using Regime Switching VAR-SV models. Unpublished Manuscript, Department of Economics, Pontificia Universidad Católica del Perú.
- [2] Arena, M., and Tuesta, P. (1999). El objetivo de la intervención del banco central: ¿el nivel del tipo de cambio, la reducción de la volatilidad cambiaria o ambos?: Un análisis de la experiencia peruana 1991-1998. Revista Estudios Económicos 5, 61-88.
- [3] Benati, L. (2008). The "Great Moderation" in the United Kingdom. Journal of Money, Credit, and Banking 40(1), 121-47. doi: https://doi.org/10.1111/j.1538-4616.2008.00106.x.
- Bernanke, B., and Mihov, I. (1998). Measuring monetary policy. The Quartely Journal of Economics 113(3), 869-902. doi: https://doi.org/10.1162/003355398555775.
- [5] Bigio, S., and Salas, J. (2006). Efectos no lineales de choques de política monetaria y de tipo de cambio real en economías parcialmente dolarizadas: un análisis empírico para el Perú. Working Paper 8, Banco Central de Reserva del Perú.
- [6] Bittencourt, M., Mwabutwa, C., and Viegi, N. (2016). Evolution of Monetary Policy Transmission Mechanism in Malawi TVP-VAR approach. *Journal of Economic Development* 41(1), 33-55. doi: https://doi.org/10.35866/caujed.2016.41.1.003.
- Boivin, J. and Giannoni, M. P. (2006). Has monetary policy become more effective?. Review of Economics and Statistics 88(3), 445-462. doi: https://doi.org/10.1162/rest.88.3.445.
- [8] Bringas, P., and Tuesta, V. (1997). El superávit de encaje y los mecanismos de transmisión de la política monetaria: una aproximación. *Revista Estudios Económicos* 1. Banco Central de Reserva del Perú.

- [9] Canova, F., Gambetti, L., and Pappa, E. (2007). The Structural Dynamics of GDP Growth and Inflation: Some International Evidence. *The Economic Journal* 117(519), 167-191. doi: https://doi.org/10.1111/j.1468-0297.2007.02040.x.
- [10] Castillo, P., and Barco, D. (2009). Crisis financieras y manejo de reservas en el Perú. Revista Estudios Económicos 17, 85-112.
- [11] Castillo, P., Montoro, C., and Tuesta, V. (2009). Dinero, inflación y tasas de interés: ¿cambia el vínculo cuando cambia el instrumento de política monetaria?. Working Paper 001, Banco Central de Reserva del Perú.
- [12] Castillo, P., Montoya, J., and Quineche, R. (2016). From the "Great Inflation" to the "Great Moderation" in Peru: A Time Varying Structural Vector Autoregressions Analysis. Working Paper 003, Banco Central de Reserva del Perú.
- [13] Castillo, P., Peréz Forero, F., and Tuesta, V. (2011). Los mecanismos de transmisión de la política monetaria en Perú. *Revista de Estudios Económicos* 21, 41-63.
- [14] Chan, J. C., and Eisenstat, E. (2015). Marginal likelihood estimation with the cross-entropy method. *Econometric Reviews* 34(3), 256-285. doi: https://doi.org/10.1080/07474938. 2014.944474.
- [15] Chan, J. C. (2022). Asymmetric conjugate priors for large Bayesian VARs. Quantitative Economics 13(3), 1145-1169. doi: https://doi.org/10.3982/QE1381.
- [16] Chan, J. C., and Yu, X. (2022). Fast and Accurate Variational Inference for Large Bayesian VARs with Stochastic Volatility. *Journal of Economic Dynamics and Control* 143. doi: https: //doi.org/10.1016/j.jedc.2022.104505.
- [17] Chan, J. C., and Eisenstat, E. (2018). Bayesian Model Comparison for Time-Varying Parameter VARs with Stochastic Volatility. *Journal of Applied Econometrics* 33(4), 509-532. doi: https://doi.org/10.1002/jae.2617.
- [18] Chan, J. C., and Grant, A. (2016). On the observed-data deviance information criterion for volatility modeling. *Journal of Financial Econometrics* 14(4), 772-802. doi: https://doi. org/10.1016/j.csda.2014.07.018.
- [19] Chan, J. C., and Jeliazkov, I. (2009). Efficient simulation and integrated likelihood estimation in state space models. *International Journal of Mathematical Modelling and Numerical Optimisation* 1(1-2), 101-120. doi: https://doi.org/10.1504/IJMMND.2009.030090.
- [20] Chávez, P., and Rodríguez, G. (2023). Time changing effects of external shocks on macroeconomic fluctuations in Peru: empirical application using regime-switching VAR models with stochastic volatility. *Review of World Economics* 159(2), 505-544. doi: https://doi.org/ 10.1007/s10290-022-00474-1.
- [21] Christiano, L., Eichenbaum, M., and Evans, C. (1999). Monetary policy shocks: What have we learned and to what end? In Taylor, J. and M. Woodford (Ed.), *Handbook of Macroeconomics*, Chapter 2 1, 65-148. doi: http://doi.org/10.1016/S1574-0048(99)01005-8.
- [22] Clark, T. E. (2011). Real-time density forecasts from Bayesian vector autoregressions with stochastic volatility. *Journal of Business and Economic Statistics* 29(3), 327-341. doi: https: //doi.org/10.1198/jbes.2010.09248.

- [23] Clark, T. E., and Ravazzolo, F. (2015). Macroeconomic forecasting performance under alternative specifications of time-varying volatility. *Journal of Applied Econometrics* **30(4)**, 551-575. doi: https://doi.org/10.1002/jae.2379.
- [24] Cogley, T., and Sargent, T. J. (2001). Evolving Post-World War II U.S. Inflation Dynamics. NBER Macroeconomic Annual 16, 331-373. doi: https://doi.org/10.1086/654451.
- [25] Cogley, T., and Sargent, T. J. (2005). Drifts and Volatilities: Monetary Policies and Outcomes in the Post WWII US. *Review of Economic Dynamics* 8, 262-302. doi: https://doi.org/10. 1016/j.red.2004.10.009.
- [26] Dancourt, O., and Jiménez, R. (2010). Perú: lecciones de la recesión de 2008-2009. MPRA Paper 74889, University Library of Munich.
- [27] Dancourt, O. (2013). Reglas de política monetaria y choques externos en una economía semidolarizada. *Economía* 36(72), 91-116.
- [28] Drechsel, T., and Tenreyro, S. (2018). Commodity booms and busts in emerging economies. Journal of International Economics 112, 200-218. doi: https://doi.org/10.1016/j. jinteco.2017.12.009.
- [29] Fernández, A., González, A., and Rodriguez, D. (2018). Sharing a ride on the commodities roller coaster: Common factors in business cycles of emerging economies. *Journal of International Economics* 111, 99-121. doi: https://doi.org/10.1016/j.jinteco.2017.11.008.
- [30] Franta, M., Horvath, R., and Rusnak, M. (2013). Evaluating Changes in the Monetary Transmission Mechanism in the Czech Republic. *Empirical Economics* 46(3), 827-842. doi: https://doi.org/10.1007/s00181-013-0699-0.
- [31] Gordon, D., and Leeper, E. (1994). The Dynamic Impacts of Monetary Policy: An Exercise in Tentative Identification. *Journal of Political Economy* 102(6), 1228-1247. doi: https: //doi.org/10.1086/261969.
- [32] Guevara, C., and Rodríguez, G. (2020). The role of credit supply shocks in pacific alliance countries: A TVP-VAR-SV approach. *The North American Journal of Economics and Finance* 52, 101140. doi: https://doi.org/10.1016/j.najef.2019.101140.
- [33] Kim, S., Shephard, N., and S. Chib. (1998). Stochastic Volatility: Likelihood Inference and Comparison with ARCH Models. *The Review of Economic Studies* 65(3), 361-393. doi: https: //doi.org/10.1111/1467-937X.00050.
- [34] Koop, G., Leon-Gonzalez, R., and Strachan, R. W. (2009). On the evolution of the monetary policy transmission mechanism. *Journal of Economic Dynamics and Control* 33, 997-1017. doi: https://doi.org/10.1016/J.JEDC.2008.11.003.
- [35] Lahura, E. (2006). El efecto traspaso de la tasa de interés y la política monetaria en el Perú: 1995-2004. *Revista Estudios Económicos* 13.
- [36] Lahura, E. (2010). The effects of monetary policy shocks in Peru: Semi-structural identification using a factor-augmented vector autoregressive model. Working Paper 8, Banco Central de Reserva del Perú.

- [37] Lahura, E. (2017). El efecto traspaso de la tasa de interés de política monetaria en Perú: Evidencia reciente. *Revista Estudios Económicos* 9.
- [38] León Fernández, D. (1999). La información contenida en los agregados monetarios en el Perú. Revista Estudios Económicos 5.
- [39] Mendoza, W. (2013). Contexto internacional y desempeño macroeconómico en América Latina y el Péru: 1980-2012. Documento de Trabajo 351. Departamento de Economía, Pontificia Universidad Católica del Perú.
- [40] Nakajima, J. (2011). Monetary Policy Transmission under Zero Interest Rates: An Extended Time-Varying Parameter Vector Autoregression Approach. The B.E. Journal of Macroeconomics 11(1), 1-24. doi: https://doi.org/10.2202/1935-1690.2323.
- [41] Ojeda Cunya, J. A. and Rodríguez, G. (2022). Time-Varying Effects of External Shocks on Macroeconomic Fluctuations in Peru: An Empirical Application using TVP-VAR-SV Models. Working Paper 507. Department of Economics, Pontificia Universidad Católica del Perú. doi: http://doi.org/10.18800/2079-8474.0507.
- [42] Pérez Forero, F. (2015). Comparing the Transmission of Monetary Policy Shocks in Latin America: A Hierarchical Panel VAR. Working Paper 15, Banco Central de Reserva del Perú.
- [43] Portilla, J., Rodríguez, G., and Castillo B., P. (2022). Evolution of Monetary Policy in Peru: An Empirical Application Using a Mixture Innovation TVP-VAR-SV Model. *CESifo Economic Studies* 68(1), 98-126. doi: https://doi.org/10.1093/cesifo/ifab013.
- [44] Primiceri, G. (2005). Time Varying Structural Vector Autoregressions and Monetary Policy. Review of Economic Studies 72, 821-852. doi: https://doi.org/10.1111/j.1467-937X. 2005.00353.x.
- [45] Quispe, Z. (1998). Una aproximación a la demanda de los principales agregados monetarios en el Perú: Junio 1991-Mayo 1997. *Revista Estudios Económicos* 3, 1-20.
- [46] Quispe, Z. (2000). Monetary policy in a dollarized economy: The case of Peru. In Mahadeva, L., and Sterne, G. (Eds.), *Monetary Policy Framewok in a Global Context* (330-353). London: Routledge and Bank of England. doi: https://doi.org/10.4324/9780203351147.
- [47] Quispe, Z. (2001). Transmission mechanisms of monetary policy in a economy with partial dollarisation: the case of Peru. In Bank for International Settlements (Ed.), Modelling aspects of the inflation process and the monetary transmission mechanism in emerging market countries (210-231). Basel: Bank for International Settlements. doi: http://dx.doi.org/10. 2139/ssrn.1187568.
- [48] Quispe, Z., Rodríguez, D., Toma, H., and Vasquez, C. (2017). Choques externos y coordinación de políticas monetarias y macroprudenciales en las economías de la Alianza del Pacífico. *Revista Estudios Económicos* 34, 31-54.
- [49] Rodríguez, G., Villanueva, P., and Castillo B., P. (2018). Driving Economic Fluctuations in Peru: The Role of the Terms of Trade. *Empirical Economics* 55, 1089-119. doi: http: //dx.doi.org/10.1007/s00181-017-1318-2.

- [50] Rodríguez, G., Castillo B., P., and Hasegawa, H. (2023a). Does the Central Bank of Peru Respond to exchange rate movements? A Bayesian estimation of a New Keynesian DSGE model with FX interventions. *The North American Journal of Economics and Finance* 68, 101965. doi: https://doi.org/10.1016/j.najef.2023.101965.
- [51] Rodríguez, G., Vassallo, R., and P. Castillo B. (2023b). Effects of external shocks on macroeconomic fluctuations in Pacific Alliance countries. *Economic Modelling* **124**, 106302. doi: https://doi.org/10.1016/j.econmod.2023.106302.
- [52] Rossini, R., and Vega, M. (2007). El mecanismo de transmisión de la política monetaria en un entorno de dolarización financiera: El caso del Perú entre 1996 y 2006. Working Paper 017, Banco Central de Reserva del Perú.
- [53] Rossini, R., Quispe, Z., and Rodríguez, D. (2013). Flujo de capitales, política monetaria e intervención cambiaria en el Perú. *Revista Estudios Económicos* **25**, 39-50.
- [54] Rossini, R., Quispe, Z., and Serrano, E. (2014). Intervención cambiaria en el Perú: 2007 a 2013. Revista Estudios Económicos 27(27), 9-24.
- [55] Rossini, R. (2016). La política monetaria del BCRP en los últimos 25 años. In Yamada, G., and Winkelried, D. (Eds.), Política y Estabilidad Monetaria en el Perú: Homenaje a Julio Velarde, Banquero Central del año 2015 (pp. 23-35). Lima: Universidad del Pacífico.
- [56] Sims, C. (1980). Macroeconomics and Reality. *Econometrica* 48(1), 1-48.
- [57] Sims, C. (2001). Comment on Sargent and Cogley's "Evolving Post-World War II U.S. Inflation Dynamics". NBER Macroeconomics Annual 16, 373-379. doi: https://doi.org/10.1086/ 654452.
- [58] Sims, C., and Zha, T. (2006). Were there regime switches in US monetary policy? American Economic Review 96(1), 54-81.
- [59] Spiegelhalter, D., Best, N., Carlin, B., and Van der Linde, A. (2002). Bayesian measures of model complexity and fit. *Journal of the Royal Statistical Society* 64(4), 583-639. doi: https://doi.org/10.1111/1467-9868.00353.
- [60] Stock, J. H. (2001). Discussion of Cogley and Sargent "Evolving Post-World War II U.S. Inflation Dynamics". NBER Macroeconomics Annual 16, 379-387. doi: https://doi.org/ 10.1086/654453.
- [61] Velarde, J., and Rodríguez, M. (2001). Efectos de la crisis financiera internacional en la economía peruana 1997-1998: lecciones e implicancias de política económica. Documento de Trabajo 36, Centro de Investigación de la Universidad del Pacífico.
- [62] Velarde, J. (2015). The Future of Monetary Policy. In Werner, A., and Santos, A. (Eds.), *Peru: Staying the Course of Economic Success* (417-464). Washington D.C.: IMF. doi: https://doi.org/10.5089/9781513599748.071.
- [63] Viladegut, H., and Cabello, M. (2014). El canal del crédito en el Perú: Una aproximación SVAR. Revista Estudios Económicos 27, 51-66.
- [64] Winkelried, D. (2003). ¿Es asimétrico el pass-through en el Perú?: Un análisis agregado. Revista Estudios Económicos 10, 1-29.

- [65] Winkelried, D. (2012). Traspaso del tipo de cambio y metas de inflación en el Perú. *Revista Estudios Económicos* 23, 9-24.
- [66] Wu, F., and Wei, Y. (2016). Research on the Impact of Monetary Policy on CPI in China based on TVP-VAR Model. International Journal of Simulation - Systems, Science and Technology 17(38), 9.1-9.6. doi: https://doi.org/10.5013/IJSSST.a.17.38.09.

	Trace	e Test										
trace	16% perc.	50% perc.	84% perc.									
0.240	0.166	0.231	0.344									
	Kolmogorov-	Smirnov Test										
	ົ	$t_t$										
1993Q4-2002Q1	$2002\mathrm{Q2}\text{-}2018\mathrm{Q2}$	1996Q3-2007Q2	$2007 \mathrm{Q2}\text{-}2018 \mathrm{Q2}$									
15/15	15/15	14/15	14/15									
	ß	$\boldsymbol{B}_t$										
1993Q4-2002Q1	$2002\mathrm{Q2}\text{-}2018\mathrm{Q2}$	1996Q3-2007Q2	$2007\mathrm{Q2}\text{-}2018\mathrm{Q2}$									
58/78	54/78	63/78	63/78									
	h	$\mathbf{h}_t$										
1993Q4-2002Q1	$2002\mathrm{Q2}\text{-}2018\mathrm{Q2}$	1996Q3-2007Q2	2007Q3-2018Q2									
6/6	6/6	6/6	6/6									
	t-test											
	ົ	$t_t$										
1993Q4-2002Q1	$2002\mathrm{Q2}\text{-}2018\mathrm{Q2}$	1996Q3-2007Q2	2007Q3-2018Q2									
15/15	15/15	15/15	15/15									
	ß	$\boldsymbol{B}_t$										
1993Q4-2002Q1	$2002\mathrm{Q2}\text{-}2018\mathrm{Q2}$	1996Q3-2007Q2	2007Q3-2018Q2									
58/78	54/78	62/78	54/78									
	h	$\mathbf{h}_t$										
1993Q4-2002Q1	2002Q2-2018Q2	1996Q3-2007Q2	2007Q3-2018Q2									
6/6	6/6	6/6	6/6									

Table 1. Tests for Time Variation in the Coefficients and Volatility

 $\boldsymbol{\gamma}_t$  represents the coefficients of contemporaneous relationships,  $\boldsymbol{\beta}_t$  are the coefficients associated to intercepts and lagged variables and  $\mathbf{h}_t$  are the variances of innovations.  $\boldsymbol{\gamma}_t$  has  $(n \times (n-1))/2$  elements which are under the diagonal,  $\boldsymbol{\beta}_t$  has n intercepts and  $n \times n \times p$  parameters related to lags, and  $\mathbf{h}_t$  has n elements.

Table 2. Models Selection

Model	$\text{Log-ML}_{CE}$	SD	Rank	DIC	SD	Rank
TVP-VAR-SV	-1506.455	0.288	5	2533.232	1.155	5
TVP-VAR	-1580.195	0.194	6	2633.436	2.949	7
TVP-VAR-R1-SV	-1483.666	0.264	1	2411.979	0.514	3
TVP-VAR-R2-SV	-1502.186	0.397	4	2498.542	1.657	4
TVP-VAR-R3-SV	-1484.715	0.282	<b>2</b>	2390.006	0.531	<b>2</b>
CVAR-SV	-1486.077	0.073	3	2346.120	6.300	1
CVAR	-1584.832	0.023	7	2562.849	0.351	6

Each Log-ML<sub>CE</sub> estimate is based on 10,000 evaluations of the integrated likelihood using 10 parallel chains, where the importance sampling density is constructed using 10,000 posterior draws after a burn-in period of 1,000. Each DIC estimate (and the corresponding numerical standard error) is computed using 10 parallel chains; each consists of 10,000 posterior draws after a burn-inperiod of 1,000. The integrated likelihood and the DIC are evaluated every 10th post burn-in draw-a total of 10,000 evaluations.

			Tr	ace Te	est							
Case	pri	or trac	е	16	% perc	•	50	% perc		849	7 perc	•
Alternative Priors	(	0.240		(	0.164			0.231		(	).338	
Alternative Order 1	(	0.240		(	0.163			0.231		(	).339	
Alternative Order 2	(	0.240		(	0.164			0.232		(	).344	
Non-Primary GDP Growth	(	0.240		(	0.165			0.233		(	).343	
EPI Growth	(	0.240		(	0.437		(	0.525		(	).642	
		Kolm	ogor	ov-Sm	irnov [	ſest						
	19960	Q3-2002	2Q1	20020	<b>Q</b> 2-201	8Q2	19960	Q3-200'	7Q2	20070	<b>)</b> 3-2018	8Q2
	$oldsymbol{\gamma}_t$	$oldsymbol{eta}_t$	$\mathbf{h}_t$									
Alternative Priors	14/15	58/78	6/6	15/15	60/78	6/6	15/15	64/78	6/6	15/15	63/78	6/6
Alternative Order 1	15/15	59/78	6/6	15/15	61/78	6/6	15/15	64/78	6/6	15/15	61/78	6/6
Alternative Order 2	13/15	56/78	6/6	14/15	61/78	6/6	14/15	64/78	6/6	14/15	62/78	6/6
Non-Primary GDP Growth	13/15	57/78	6/6	15/15	63/78	6/6	14/15	65/78	6/6	15/15	61/78	6/6
EPI Growth	15/15	60/78	6/6	14/15	70/78	6/6	15/15	68/78	6/6	15/15	64/78	6/6
				t-test								
	19960	23-2002	2Q1	20020	Q2-201	8Q2	19960	Q3-200'	7Q2	20070	23-2018	8Q2
	$oldsymbol{\gamma}_t$	$oldsymbol{eta}_t$	$\mathbf{h}_t$									
Alternative Priors	15/15	61/78	6/6	15/15	56/78	6/6	15/15	62/78	6/6	15/15	55/78	6/6
Alternative Order 1	15/15	56/78	6/6	15/15	55/78	6/6	15/15	61/78	6/6	15/15	54/78	6/6
Alternative Order 2	15/15	55/78	6/6	15/15	55/78	6/6	15/15	61/78	6/6	15/15	58/78	6/6
Non-Primary GDP Growth	15/15	54/78	6/6	15/15	57/78	6/6	15/15	65/78	6/6	15/15	59/78	6/6
EPI Growth	15/15	57/78	6/6	15/15	67/78	6/6	15/15	64/78	6/6	15/15	60/78	6/6

Table 3. Robustness Analysis: Tests for Time Variation in the Coefficients and Volatility

 $\gamma_t$  represents the coefficients of contemporaneous relationships,  $\beta_t$  are the coefficients associate to intercepts and lagged variables and  $\mathbf{h}_t$  are the variances of innovations.  $\gamma_t$  has  $(n \times (n-1))/2$  elements which are under the diagonal,  $\beta_t$  has n intercepts and  $n \times n \times p$  parameters related to lags, and  $\mathbf{h}_t$  has n elements. Alternative Order 1 refers to a change in the order between Money Supply Growth and Inflation, while Alternative Order 2 refers to a change between GPD Growth and Inflation.

Table 4. Robustness Analysis: Models Selection

	Rank	ы		7		3		4		7		1		9	
rowth	DIC	2353.377	(1.587)	2462.609	(5.941)	2279.463	(0.570)	2326.509	(1.700)	2251.786	(0.635)	2174.346	(9.396)	2434.973	(0.239)
EPI C	Rank	ъ		9		S		4		1		2		7	
(	$\operatorname{Log-ML} CE$	-1417.976	(0.288)	-1502.660	(0.553)	-1412.237	(0.269)	-1413.504	(0.342)	-1411.613	(0.389)	-1411.865	(0.044)	-1508.106	(0.017)
wth	Rank	ъ		7		33		4		2		1		9	
GDP Gro	DIC	2512.303	(1.922)	2631.564	(7.765)	2402.684	(0.765)	2485.108	(1.783)	2375.287	(0.599)	2312.340	(6.333)	2571.306	(0.312)
imary	Rank	ъ		9		7		4		1		e		7	
(iv) Non-P <sub>1</sub>	$\mathrm{Log}\text{-}\mathrm{ML}CE$	-1498.527	(0.225)	-1585.301	(0.778)	-1481.089	(0.306)	-1495.687	(0.331)	-1479.541	(0.255)	-1482.588	(0.078)	-1588.385	(0.019)
	Rank	ъ		7		3		4		7		1		9	
ve Order 2	DIC	2534.292	(1.717)	2635.119	(2.269)	2409.227	(1.110)	2499.850	(1.148)	2389.535	(0.416)	2338.071	(6.454)	2563.461	(0.198)
ernati	Rank	ىد م		7		1		4		2		c,		9	
(iii) Alt	$\operatorname{Log-ML}_{CE}$	-1505.273	(0.373)	-1577.118	(0.305)	-1483.021	(0.321)	-1500.747	(0.626)	-1483.884	(0.218)	-1484.379	(0.061)	-1583.421	(0.024)
	Rank	ы		2		e S		4		2		1		9	
/e Order 1	DIC	2534.125	(1.808)	2640.373	(6.725)	2417.232	(0.679)	2501.206	(1.849)	2392.546	(0.679)	2344.169	(8.139)	2563.336	(0.363)
ernativ	Rank	ъ		2		1		4		2		c,		9	
(ii) Alte	$\operatorname{Log-ML}_{CE}$	-1508.676	(0.249)	-1589.155	(0.723)	-1487.594	(0.219)	-1505.562	(0.250)	-1488.263	(0.319)	-1488.696	(0.045)	-1587.382	(0.019)
	Rank	n		7		3		4		7		1		9	
ive Priors	DIC	2536.125	(1.808)	2645.197	(1.602)	2416.764	(0.710)	2505.206	(1.849)	2398.942	(1.151)	2346.169	(8.139)	2567.320	(0.202)
ernati	Rank	ىد م		9		1		4		2		e		7	
(i) Alt	$\operatorname{Log-ML}_{CE}$	-1536.295	(0.210)	-1572.295	(0.210)	-1476.532	(0.315)	-1520.295	(0.210)	-1479.293	(0.318)	-1480.271	(0.104)	-1574.124	(0.022)
	Model	TVP-VAR-SV		TVP-VAR		TVP-VAR-R1-SV		TVP-VAR-R2-SV		TVP-VAR-R3-SV		CVAR-SV		CVAR	

using 10,000 posterior draws after a burn-in period of 1,000. Each DIC estimate (and the corresponding numerical standard error) is computed using 10 parallel chains; each consists of 10,000 posterior draws after a burn-inperiod of 1,000. The integrated likelihood and the DIC are evaluated every 10th post burn-in draw-a total of 10,000 evaluations. Alternative Order 1 refers to a change between Money Supply Growth and Inflation in the order, while Alternative Order 2 refers to Each Log-ML $_{CE}$  estimate is based on 10,000 evaluations of the integrated likelihood using 10 parallel chains, where the importance sampling density is constructed a change between GPD Growth and Inflation.













2000<sup>2005</sup>2010

S

10 15

2000<sup>2005</sup>2010<sup>2015</sup>

2

10

2000<sup>2005</sup>2010<sup>2015</sup>

2 10

2000<sup>2005</sup>2010

ß 10

15

20

15

20

-0.10

-0.10

-0.05

15

20

20

-0.05 -0.10

-0.05 -0.10





















# ÚLTIMAS PUBLICACIONES DE LOS PROFESORES DEL DEPARTAMENTO DE ECONOMÍA

Libros

Luis García 2023 *Econometría 1. Tercera edición.* Lima, Fondo Editorial PUCP.

Waldo Mendoza 2023 *Constitución y crecimiento económico: Perú 1993-2021*. Lima, Fondo Editorial PUCP.

Oscar Dancourt y Waldo Mendoza (Eds.) 2023 *Ensayos macroeconómicos en honor a Félix Jiménez*. Lima, Fondo Editorial PUCP.

Carlos Contreras (Editor)

2022 Historia económica del Perú central. Historia económica del Perú central. Ventajas y desafíos de estar cerca de la capital.

Alejandro Lugon

2022 Equilibrio, eficiencia e imperfecciones del mercado. Lima, Fondo Editorial PUCP.

## Waldo Mendoza Bellido

2022 Cómo investigan los economistas. Guía para elaborar y desarrollar un proyecto de investigación. Segunda edición aumentada. Lima, Fondo Editorial PUCP.

Elena Álvarez (Editor)

2022 Agricultura y desarrollo rural en el Perú: homenaje a José María Caballero. Lima, Departamento de Economía PUCP.

Aleida Azamar Alonso, José Carlos Silva Macher y Federico Zuberman (Editores)

2022 *Economía ecológica latinoamericana.* Buenos Aires, México. CLACSO, Siglo XXI Editores.

Efraín Gonzales de Olarte

2021 Economía regional y urbana. El espacio importa. Lima, Fondo Editorial PUCP.

# Alfredo Dammert Lira

2021 *Economía minera*. Lima, Fondo Editorial PUCP.

# Adolfo Figueroa

2021 The Quality of Society, Volume II – Essays on the Unified Theory of Capitalism. New York, Palgrave Macmillan.

Carlos Contreras Carranza (Editor)

2021 La Economía como Ciencia Social en el Perú. Cincuenta años de estudios económicos en la Pontificia Universidad Católica del Perú. Lima, Departamento de Economía PUCP.

José Carlos Orihuela y César Contreras

2021 Amazonía en cifras: Recursos naturales, cambio climático y desigualdades. Lima, OXFAM. Alan Fairlie

2021 *Hacia una estrategia de desarrollo sostenible para el Perú del Bicentenario.* Arequipa, Editorial UNSA.

### Waldo Mendoza e Yuliño Anastacio

2021 La historia fiscal del Perú: 1980-2020. Colapso, estabilización, consolidación y el golpe de la COVID-19. Lima, Fondo Editorial PUCP.

Cecilia Garavito

2020 *Microeconomía: Consumidores, productores y estructuras de mercado. Segunda edición.* Lima, Fondo Editorial de la Pontificia Universidad Católica del Perú.

Adolfo Figueroa

2019 *The Quality of Society Essays on the Unified Theory of Capitalism.* New York. Palgrave MacMillan.

Carlos Contreras y Stephan Gruber (Eds.)

2019 *Historia del Pensamiento Económico en el Perú. Antología y selección de textos.* Lima, Facultad de Ciencias Sociales PUCP.

Barreix, Alberto Daniel; Corrales, Luis Fernando; Benitez, Juan Carlos; Garcimartín, Carlos; Ardanaz, Martín; Díaz, Santiago; Cerda, Rodrigo; Larraín B., Felipe; Revilla, Ernesto; Acevedo, Carlos; Peña, Santiago; Agüero, Emmanuel; Mendoza Bellido, Waldo; Escobar Arango y Andrés.

2019 Reglas fiscales resilientes en América Latina. Washington, BID.

José D. Gallardo Ku

2019 *Notas de teoría para para la incertidumbre*. Lima, Fondo Editorial de la Pontificia Universidad Católica del Perú.

Úrsula Aldana, Jhonatan Clausen, Angelo Cozzubo, Carolina Trivelli, Carlos Urrutia y Johanna Yancari

2018 *Desigualdad y pobreza en un contexto de crecimiento económico*. Lima, Instituto de Estudios Peruanos.

Séverine Deneulin, Jhonatan Clausen y Arelí Valencia (Eds.)

2018 Introducción al enfoque de las capacidades: Aportes para el Desarrollo Humano en América Latina. Flacso Argentina y Editorial Manantial. Fondo Editorial de la Pontificia Universidad Católica del Perú.

Mario Dammil, Oscar Dancourt y Roberto Frenkel (Eds.)

2018 *Dilemas de las políticas cambiarias y monetarias en América Latina*. Lima, Fondo Editorial de la Pontificia Universidad Católica del Perú.

# Documentos de trabajo

- No. 522 Perú 1990-2021: la causa del "milagro" económico ¿Constitución de 1993 o Superciclo de las materias primas?
   Félix Jiménez, José Oscátegui y Marco Arroyo. Agosto, 2023.
- No. 521 Envejeciendo desigualmente en América Latina. Javier Olivera. Julio, 2023.
- No. 520 Choques externos en la economía peruana: un enfoque de ceros y signos en un modelo BVAR.
  Gustavo Ganiko y Álvaro Jiménez. Mayo, 2023
- No. 519 Ley de Okun en Lima Metropolitana 1970 2021. Cecilia Garavito. Mayo, 2023
- No. 518 Efectos 'Spillovers' (de derrame) del COVID-19 Sobre la Pobreza en el Perú: Un Diseño No Experimental de Control Sintético. Mario Tello. Febrero, 2023
- No. 517 Indicadores comerciales de la Comunidad Andina 2002-2021: ¿Posible complementariedad o convergencia regional? Alan Fairlie y Paula Paredes. Febrero, 2023.
- No. 516 Evolution over Time of the Effects of Fiscal Shocks in the Peruvian Economy: Empirical Application Using TVP-VAR-SV Models. Alexander Meléndez Holguín y Gabriel Rodríguez. Enero, 2023.
- No. 515 COVID-19 and Gender Differences in the Labor Market: Evidence from the Peruvian Economy. Giannina Vaccaro, Tania Paredes. Julio 2022.
- No. 514 Do institutions mitigate the uncertainty effect on sovereign credit ratings? Nelson Ramírez-Rondán, Renato Rojas-Rojas y Julio A. Villavicencio. Julio 2022.
- No. 513 Gender gap in pension savings: Evidence from Peru's individual capitalization system. Javier Olivera y Yadiraah Iparraguirre. Junio 2022.
- No. 512 Poder de mercado, bienestar social y eficiencia en la industria microfinanciera regulada en el Perú. Giovanna Aguilar y Jhonatan Portilla. Junio 2022.
- No. 511 Perú 1990-2020: Heterogeneidad estructural y regímenes económicos regionales ¿Persiste la desconexión entre la economía, la demografía y la geografía? Félix Jiménez y Marco Arroyo. Junio 2022.
- No. 510 Evolution of the Exchange Rate Pass-Throught into Prices in Peru: An Empirical Application Using TVP-VAR-SV Models. Roberto Calero, Gabriel Rodríguez y Rodrigo Salcedo Cisneros. Mayo 2022.
- No. 509 Time Changing Effects of External Shocks on Macroeconomic Fluctuations in Peru: Empirical Application Using Regime-Switching VAR Models with Stochastic Volatility. Paulo Chávez y Gabriel Rodríguez. Marzo 2022.

- No. 508 Time Evolution of External Shocks on Macroeconomic Fluctuations in Pacific Alliance Countries: Empirical Application using TVP-VAR-SV Models. Gabriel Rodríguez y Renato Vassallo. Marzo 2022.
- No. 507 Time-Varying Effects of External Shocks on Macroeconomic Fluctuations in Peru: An Empirical Application using TVP-VARSV Models. Junior A. Ojeda Cunya y Gabriel Rodríguez. Marzo 2022.
- No. 506 La Macroeconomía de la cuarentena: Un modelo de dos sectores. Waldo Mendoza, Luis Mancilla y Rafael Velarde. Febrero 2022.
- No. 505 ¿Coexistencia o canibalismo? Un análisis del desplazamiento de medios de comunicación tradicionales y modernos en los adultos mayores para el caso latinoamericano: Argentina, Colombia, Ecuador, Guatemala, Paraguay y Perú. Roxana Barrantes Cáceres y Silvana Manrique Romero. Enero 2022.
- No. 504 "Does the Central Bank of Peru Respond to Exchange Rate Movements? A Bayesian Estimation of a New Keynesian DSGE Model with FX Interventions". Gabriel Rodríguez, Paul Castillo B. y Harumi Hasegawa. Diciembre, 2021
- No. 503 "La no linealidad en la relación entre la competencia y la sostenibilidad financiera y alcance social de las instituciones microfinancieras reguladas en el Perú".
  Giovanna Aguilar y Jhonatan Portilla. Noviembre, 2021.
- No. 502 "Approximate Bayesian Estimation of Stochastic Volatility in Mean Models using Hidden Markov Models: Empirical Evidence from Stock Latin American Markets". Carlos A. Abanto-Valle, Gabriel Rodríguez, Luis M. Castro Cepero y Hernán B. Garrafa-Aragón. Noviembre, 2021.
- No. 501 "El impacto de políticas diferenciadas de cuarentena sobre la mortalidad por COVID-19: el caso de Brasil y Perú". Angelo Cozzubo, Javier Herrera, Mireille Razafindrakoto y François Roubaud. Octubre, 2021.
- No. 500 "Determinantes del gasto de bolsillo en salud en el Perú". Luis García y Crissy Rojas. Julio, 2021.
- No. 499 "Cadenas Globales de Valor de Exportación de los Países de la Comunidad Andina 2000-2015". Mario Tello. Junio, 2021.
- No. 498 "¿Cómo afecta el desempleo regional a los salarios en el área urbana? Una curva de salarios para Perú (2012-2019)". Sergio Quispe. Mayo, 2021.
- No. 497 "¿Qué tan rígidos son los precios en línea? Evidencia para Perú usando Big Data". Hilary Coronado, Erick Lahura y Marco Vega. Mayo, 2021.
- No. 496 "Reformando el sistema de pensiones en Perú: costo fiscal, nivel de pensiones, brecha de género y desigualdad". Javier Olivera. Diciembre, 2020.
- No. 495 "Crónica de la economía peruana en tiempos de pandemia". Jorge Vega Castro. Diciembre, 2020.
- No. 494 "Epidemia y nivel de actividad económica: un modelo". Waldo Mendoza e Isaías Chalco. Setiembre, 2020.

- No. 493 "Competencia, alcance social y sostenibilidad financiera en las microfinanzas reguladas peruanas". Giovanna Aguilar Andía y Jhonatan Portilla Goicochea. Setiembre, 2020.
- No. 492 "Empoderamiento de la mujer y demanda por servicios de salud preventivos y de salud reproductiva en el Perú 2015-2018". Pedro Francke y Diego Quispe O. Julio, 2020.
- No. 491 "Inversión en infraestructura y demanda turística: una aplicación del enfoque de control sintético para el caso Kuéalp, Perú". Erick Lahura y Rosario Sabrera. Julio, 2020.
- No. 490 "La dinámica de inversión privada. El modelo del acelerador flexible en una economía abierta". Waldo Mendoza Bellido. Mayo, 2020.
- No. 489 "Time-Varying Impact of Fiscal Shocks over GDP Growth in Peru: An Empirical Application using Hybrid TVP-VAR-SV Models". Álvaro Jiménez y Gabriel Rodríguez. Abril, 2020.
- No. 488 "Experimentos clásicos de economía. Evidencia de laboratorio de Perú". Kristian López Vargas y Alejandro Lugon. Marzo, 2020.
- No. 487 "Investigación y desarrollo, tecnologías de información y comunicación e impactos sobre el proceso de innovación y la productividad". Mario D. Tello. Marzo, 2020.
- No. 486 "The Political Economy Approach of Trade Barriers: The Case of Peruvian's Trade Liberalization". Mario D. Tello. Marzo, 2020.
- No. 485 "Evolution of Monetary Policy in Peru. An Empirical Application Using a Mixture Innovation TVP-VAR-SV Model". Jhonatan Portilla Goicochea y Gabriel Rodríguez. Febrero, 2020.
- No. 484 "Modeling the Volatility of Returns on Commodities: An Application and Empirical Comparison of GARCH and SV Models". Jean Pierre Fernández Prada Saucedo y Gabriel Rodríguez. Febrero, 2020.
- No. 483 "Macroeconomic Effects of Loan Supply Shocks: Empirical Evidence". Jefferson Martínez y Gabriel Rodríguez. Febrero, 2020.
- No. 482 "Acerca de la relación entre el gasto público por alumno y los retornos a la educación en el Perú: un análisis por cohortes". Luis García y Sara Sánchez. Febrero, 2020.
- No. 481 "Stochastic Volatility in Mean. Empirical Evidence from Stock Latin American Markets". Carlos A. Abanto-Valle, Gabriel Rodríguez y Hernán B. Garrafa-Aragón. Febrero, 2020.
- No. 480 "Presidential Approval in Peru: An Empirical Analysis Using a Fractionally Cointegrated VAR2". Alexander Boca Saravia y Gabriel Rodríguez. Diciembre, 2019.
- No. 479 "La Ley de Okun en el Perú: Lima Metropolitana 1971 2016." Cecilia Garavito. Agosto, 2019.

- No. 478 "Peru's Regional Growth and Convergence in 1979-2017: An Empirical Spatial Panel Data Analysis". Juan Palomino y Gabriel Rodríguez. Marzo, 2019.
- Materiales de Enseñanza
  - No. 10 "Boleta o factura: el impuesto general a las ventas (IGV) en el Perú". Jorge Vega Castro. Abril, 2023
  - No. 9 "Economía Pública. Segunda edición". Roxana Barrantes Cáceres, Silvana Manrique Romero y Carla Glave Barrantes. Febrero, 2023.
  - No. 8 "Economía Experimental Aplicada. Programación de experimentos con oTree". Ricardo Huamán-Aguilar. Febrero, 2023
  - No. 7 "Modelos de Ecuaciones Simultáneas (MES): Aplicación al mercado monetario". Luis Mancilla, Tania Paredes y Juan León. Agosto, 2022
  - No. 6 "Apuntes de Macroeconomía Intermedia". Felix Jiménez. Diciembre, 2020
  - No. 5 "Matemáticas para Economistas 1". Tessy Váquez Baos. Abril, 2019.
  - No. 4 "Teoría de la Regulación". Roxana Barrantes. Marzo, 2019.
  - No. 3 "Economía Pública". Roxana Barrantes, Silvana Manrique y Carla Glave. Marzo, 2018.
  - No. 2 "Macroeconomía: Enfoques y modelos. Ejercicios resueltos". Felix Jiménez. Marzo, 2016.
  - No. 1 "Introducción a la teoría del Equilibrio General". Alejandro Lugon. Octubre, 2015.

Departamento de Economía - Pontificia Universidad Católica del Perú Av. Universitaria 1801, San Miguel, 15008 – Perú Telf. 626-2000 anexos 4950 – 4951 <u>http://departamento.pucp.edu.pe/economia/</u>